UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT
Pursuant to Section 13 or 15(d)
of the Securities Exchange Act of 1934

Date of Report (date of earliest event reported): February 17, 2017

Matador Resources Company

(Exact name of registrant as specified in its charter)

Texas (State or Other Jurisdiction of Incorporation or Organization) 001-35410 (Commission File Number) 27-4662601 (I.R.S. Employer Identification No.)

5400 LBJ Freeway, Suite 1500, Dallas, Texas (Address of Principal Executive Offices)

75240 (Zip Code)

Registrant's telephone number, including area code: (972) 371-5200

Not Applicable (Former name or former address, if changed since last report)

Check the rovisions:	appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01. Entry into a Material Definitive Agreement.

Contribution Agreement

On February 17, 2017, in connection with the formation of San Mateo Midstream, LLC ("San Mateo Midstream"), a joint venture with FP MMP Holdings LLC, a subsidiary of Five Point Capital Partners LLC ("Five Point"), Longwood Midstream Holdings, LLC ("Longwood Holdings"), a whollyowned subsidiary of Matador Resources Company ("Matador"), entered into and consummated the transactions (the "Closing") contemplated by a Subscription and Contribution Agreement (the "Contribution Agreement"), by and among Longwood Holdings, Five Point and San Mateo Midstream.

Pursuant to the terms of the Contribution Agreement, Longwood Holdings made an initial capital contribution to San Mateo Midstream consisting of (a) all of the outstanding equity interests in Longwood RB Pipeline, LLC ("Longwood RB"), Longwood Wolf Pipeline, LLC ("Longwood Wolf"), Fulcrum Delaware Water Resources, LLC ("Fulcrum"), DLK Black River Midstream, LLC ("DLK") and Black River Water Management Company, LLC ("Black River" and, together with Longwood RB, Longwood Wolf, Fulcrum and DLK, the "Contributed Entities"), which Contributed Entities hold Matador's midstream assets in the Wolf and Rustler Breaks areas valued at \$350.0 million (the "Midstream Assets") and (b) cash in the amount of \$5.1 million, in exchange for \$1.0% of the membership interests in San Mateo Midstream. Also at the Closing, Five Point made an initial capital contribution of cash in the amount of \$176.4 million, of which \$171.5 million was immediately distributed to Longwood Holdings, partially as a reimbursement of capital expenditures by Matador and its subsidiaries with respect to the Midstream Assets, in exchange for 49.0% of the membership interests in San Mateo Midstream. In addition to the initial capital contributions, each of Longwood Holdings and Five Point committed to make additional capital contributions of \$71.4 million and \$68.6 million, respectively. Within 120 days following the Closing, the initial capital contributions will be subject to customary adjustments based on the net working capital of the Contributed Entities at the Closing and the net operating income of the Contributed Entities between February 1, 2017 and the Closing.

The Contribution Agreement contains customary representations and warranties and covenants by Longwood Holdings, Five Point and San Mateo Midstream, including covenants regarding certain tax and confidentiality matters.

The description of the Contribution Agreement set forth above is qualified in its entirety by reference to the Contribution Agreement, a copy of which is filed as Exhibit 2.1 to this Current Report on Form 8-K (this "Current Report") and is incorporated herein by reference. This summary of the principal terms and the Contribution Agreement have been included to provide security holders with information regarding the terms of the Contribution Agreement. Neither is intended to provide any other factual information about Matador, Longwood Holdings, Five Point or their respective subsidiaries or affiliates. The representations and warranties and covenants contained in the Contribution Agreement were made solely for purposes of the Contribution Agreement as of specific dates, were solely for the benefit of the parties to the Contribution Agreement, may be subject to limitations agreed upon by the contracting parties, including being qualified by confidential disclosures made for the purposes

of allocating contractual risk between the parties to the Contribution Agreement instead of establishing these matters as facts, and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to security holders. Security holders are not third-party beneficiaries under the Contribution Agreement and should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of Matador, Longwood Holdings or Five Point. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the Contribution Agreement, which subsequent information may or may not be fully reflected in Matador's public disclosures.

Supplemental Indenture

On February 17, 2017, in connection with the transactions contemplated by the Contribution Agreement (the "Contribution Transactions"), Matador entered into a Fourth Supplemental Indenture (the "Fourth Supplemental Indenture") with Wells Fargo Bank, National Association, as trustee (the "Trustee"), which supplements the Indenture, dated as of April 14, 2015 (as supplemented and amended, the "Indenture"), among Matador, the Guarantors named therein and the Trustee. Pursuant to the Fourth Supplemental Indenture, on February 17, 2017, (i) Longwood Holdings was designated as a guarantor under the Indenture and (ii) DLK and Black River were released as parties to and as guarantors under the Indenture.

The foregoing description of the Fourth Supplemental Indenture does not purport to be complete and is qualified in its entirety by reference to the Fourth Supplemental Indenture, which is attached hereto as Exhibit 4.1 and is incorporated herein by reference.

Item 2.01. Completion of Acquisition or Disposition of Assets.

On February 17, 2017, Longwood Holdings consummated the transactions contemplated by the Contribution Agreement (the "Contribution Transactions") described in Item 1.01 of this Current Report, which is incorporated by reference into this item in its entirety.

Item 7.01. Regulation FD Disclosure.

On February 17, 2017, Matador issued a press release announcing the execution of the Contribution Agreement and the Closing of the Contribution Transactions. A copy of the press release is furnished as Exhibit 99.1 to this Current Report.

The information furnished pursuant to this Item 7.01, including Exhibit 99.1, shall not be deemed to be "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, and will not be incorporated by reference into any filing under the Securities Act of 1933, as amended, unless specifically identified therein as being incorporated therein by reference.

Item 9.01. Financial Statements and Exhibits.

(b) **Pro forma financial information.**

The proforma financial information for the Contribution Transactions required by Item 9.01 is furnished as Exhibit 99.2 to this Current Report. The unaudited proforma condensed consolidated balance sheet as of September 30, 2016 has been prepared to present Matador's financial condition as if the Contribution Transactions had occurred on September 30, 2016. The unaudited proforma condensed consolidated statements of operations for the nine months ended September 30, 2016 and the fiscal year ended December 31, 2015 have been prepared to present Matador's results of operations as if the Contribution Transactions had occurred on January 1, 2016 or January 1, 2015, respectively. The unaudited proforma financial information has been prepared for comparative purposes only and does not purport to be indicative of future results of operations or financial condition.

(d) Exhibits.

Exhibit

No.	<u>Description</u>
2.1*	Subscription and Contribution Agreement, dated as of February 17, 2017, by and among Longwood Midstream Holdings, LLC, FP MMP Holdings LLC and San Mateo Midstream, LLC.
4.1	Fourth Supplemental Indenture, dated as of February 17, 2017, by and among Matador Resources Company, Black River Water Management Company, LLC, DLK Black River Midstream, LLC, Longwood Midstream Holdings, LLC, the Guarantors named therein and Wells Fargo Bank, National Association, as trustee.
99.1	Press Release, dated February 17, 2017.
99.2	Unaudited Pro Forma Condensed Consolidated Financial Information.

^{*} Pursuant to Item 601(b)(2) of Regulation S-K, Matador agrees to furnish supplementally a copy of any omitted exhibit or schedule to the SEC upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: February 24, 2017

MATADOR RESOURCES COMPANY

By: /s/ Craig N. Adams

Craig N. Adams Executive Vice President

INDEX TO EXHIBITS

Exhibit No.	<u>Description</u>		
2.1*	Subscription and Contribution Agreement, dated as of February 17, 2017, by and among Longwood Midstream Holdings, LLC, FP MMP Holdings LLC and San Mateo Midstream, LLC.		
4.1	Fourth Supplemental Indenture, dated as of February 17, 2017, by and among Matador Resources Company, Black River Water Management Company, LLC, DLK Black River Midstream, LLC, Longwood Midstream Holdings, LLC, the Guarantors named therein and Wells Fargo Bank, National Association, as trustee.		
99.1	Press Release, dated February 17, 2017.		
99.2	Unaudited Pro Forma Condensed Consolidated Financial Information.		

^{*} Pursuant to Item 601(b)(2) of Regulation S-K, the Matador agrees to furnish supplementally a copy of any omitted exhibit or schedule to the SEC upon request.

SUBSCRIPTION AND CONTRIBUTION AGREEMENT

by and among

SAN MATEO MIDSTREAM, LLC,

LONGWOOD MIDSTREAM HOLDINGS, LLC

and

FP MMP HOLDINGS LLC

Dated as of February 17, 2017

TABLE OF CONTENTS

		rage
ARTICLE I DEFINITIONS		
1.1	Certain Definitions.	1
1.2	Other Definitional and Interpretive Matters	12
ARTICLE II PURCHASE AND SALE OF THE SECURITIES		13
2.1	Purchase and Sale of the Securities.	13
2.2	Consideration.	13
2.3	Contribution Adjustment Amount.	13
ARTICLE I	II CLOSING; CLOSING DELIVERABLES	15
3.1	Closing.	15
3.2	Closing Deliverables.	15
ARTICLE I	V REPRESENTATIONS AND WARRANTIES OF LONGWOOD	16
4.1	Organization and Good Standing	16
4.2	Authorization of Agreement	17
4.3	Conflicts; Consents of Third Parties.	17
4.4	Title to the Contributed Interests; Capitalization.	18
4.5	Financial Statements and Related Information.	19
4.6	Absence of Certain Developments	19
4.7	Taxes.	19
4.8	Real Property.	21
4.9	Midstream Assets.	21
4.10	Intellectual Property.	22
4.11	Material Contracts.	22
4.12	<u>Labor; Employee Matters; Employee Benefit Plans</u> .	23
4.13	<u>Litigation</u>	25
4.14	Compliance with Laws; Permits.	25
4.15	Environmental Matters	26
4.16	<u>Insurance</u>	26
4.17	Related Party Transactions	26
4.18	Gas Regulatory Matters	27

4.40	T	
4.19	<u>Financial Advisors</u>	27
4.20	Financing; Financial Capability.	27
4.21	Investment Intent; Investment Experience	27
4.22	<u>Solvency</u>	27
4.23	Independent Investigation	27
4.24	No Other Representations or Warranties; Schedules	28
ARTICLE V REPRESENTATIONS AND WARRANTIES OF FIVE POINT		
5.1	Organization and Good Standing	28
5.2	Authorization of Agreement	28
5.3	Conflicts; Consents of Third Parties.	29
5.4	Litigation	29
5.5	Financial Advisors	29
5.6	Financing; Financial Capability.	29
5.7	Investment Intent; Investment Experience	30
5.8	<u>Independent Investigation</u>	30
5.9	No Other Representations or Warranties; Schedules	30
ARTICLE VI REPRESENTATIONS AND WARRANTIES OF THE COMPANY		31
6.1	Organization and Good Standing	31
6.2	Authorization of Agreement	31
6.3	Conflicts; Consents of Third Parties.	31
6.4	Financial Advisors	31
ARTICLE VII COVENANTS		32
7.1	Further Assurances	32
7.2	Tax Matters.	32
7.3	Confidentiality.	34
7.4	Preservation of Records.	35
7.5	<u>Publicity</u>	35
7.6	Consents	35
7.7	Additional Post-Closing Matters	36
ARTICLE VIII SURVIVAL; INDEMNIFICATION		

36

8.1

Survival.

8.2 8.3 8.4	Indemnification by the Purchasers Indemnification by the Company. Indemnification Claim Procedures.	36 37 37
8.5	<u>Limitations</u> .	39
ARTICLE IX MISCELLANEOUS		40
9.1	<u>Expenses</u>	40
9.2	Submission to Jurisdiction; Venue; Arbitration.	40
9.3	Entire Agreement; Amendments and Waivers.	42
9.4	Governing Law	43
9.5	<u>Notices</u>	43
9.6	<u>Severability</u>	44
9.7	Binding Effect; Assignment	44
9.8	Non-Recourse	44
9.9	<u>Counterparts</u> .	45

Exhibits

Rustler Breaks Assets

Exhibit A-1 Exhibit A-2 Rustler Breaks Plant Description

Exhibit A-3 Wolf Assets

Assignment and Assumption Agreement Contribution Adjustment Illustration CapEx Budget Exhibit B Exhibit C

Exhibit D

SUBSCRIPTION AND CONTRIBUTION AGREEMENT

This SUBSCRIPTION AND CONTRIBUTION AGREEMENT (this "<u>Agreement</u>"), dated as of February 17, 2017, is by and among San Mateo Midstream, LLC, a Texas limited liability company (the "<u>Company</u>"), Longwood Midstream Holdings, LLC, a Texas limited liability company ("<u>Longwood</u>"), and FP MMP Holdings LLC, a Delaware limited liability company ("<u>Five Point</u>" and, together with Longwood, the "<u>Purchasers</u>"). Capitalized terms used herein without definition shall have the respective meanings assigned to them in <u>Section 1.1</u>.

WITNESSETH:

WHEREAS, the Company has been formed as a Texas limited liability company through the filing of its certificate of formation in the office of the Secretary of State of the State of Texas;

WHEREAS, the Purchasers desire to be admitted as Members of the Company and, in connection therewith, (i) Longwood is willing to contribute the Contributed Interests and \$5,100,000 in cash to the Company and commit to contribute up to an additional \$71,400,000 in cash to the Company and (ii) Five Point is willing to contribute \$176,400,000 in cash to the Company and commit to contribute up to an additional \$68,600,000 in cash to the Company;

WHEREAS, the Company is willing (i) to acquire and accept the Contributed Interests and the Longwood Initial Cash Contribution from Longwood and (ii) to accept the Five Point Initial Cash Contribution in cash from Five Point; and

WHEREAS, in consideration of the contributions to be made by the Purchasers as described above, the Company will issue to the Purchasers the Securities, and the Purchasers shall be admitted as Members of the Company, upon the terms and subject to the conditions set forth herein and in the Limited Liability Company Agreement of San Mateo Midstream, LLC, to be entered into as of the Closing Date by and among the Purchasers (the "<u>Company Agreement</u>").

NOW, THEREFORE, in consideration of the premises and the mutual representations, warranties, covenants and agreements hereinafter contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

ARTICLE I DEFINITIONS

1.1 Certain Definitions.

For purposes of this Agreement, the following terms shall have the meanings specified in this <u>Section 1.1</u>:

"AAA" has the meaning set forth in Section 9.2(b).

- "Affiliate" means, with respect to any Person, any other Person that, directly or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such Person, and the term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities, by contract or otherwise. For purposes of this Agreement, the Company will not be deemed to be an Affiliate of Longwood or Five Point.
 - "Agreement" has the meaning set forth in the Preamble.
 - "Allocable Expenses" has the meaning set forth in the Company Agreement.
 - "Assignment and Assumption Agreement" has the meaning set forth in Section 3.2(a)(ii).
 - "Balance Sheets" has the meaning set forth in Section 4.5(a).
- "Benefit Plan" means, with respect to any Person, any "pension plan" (as defined in Section 3(2) of ERISA), any "employee benefit plan" (as defined in Section 3(3) of ERISA) and each other retirement plan, pension plan or other forms of post-retirement compensation, welfare benefit plan, employment agreement, offer letter, insurance coverage, workers compensation, severance benefits, deferred compensation, bonus plans (whether payable in cash or equity, and including annual, performance-based, change in control, retention and stay bonuses), profit sharing plan, equity or equity-based incentive compensation plan (including stock options, stock purchase, phantom stock, restricted stock, and stock appreciation rights), holiday pay, vacation, paid time off, medical, dental, vision, disability, death benefits, sick leave, fringe benefits, supplemental unemployment, personnel policy or any other similar plan, program, agreement, arrangement or policy, in each case, (i) sponsored, maintained, entered into, contributed to or required to be contributed to by such Person or an ERISA Affiliate of that Person for the benefit of any current or former director, manager, officer or employee of that Person, or (ii) with respect to which such Person may have any liability.
 - "Black River Water" means Black River Water Management Company, LLC, a Texas limited liability company and Subsidiary of Longwood.
 - "Book Basis" has the meaning set forth in the Company Agreement.
- "Books and Records" means all land, title, engineering, environmental, employee, operational and pipeline integrity data and records, customer files and documents (including credit information), supplier lists, records, literature and correspondence and other data, files, documents, instruments, Tax Returns and related workpapers and records from accountants, whether in physical or digital format, which relate to the assets and properties of the Contributed Entities (including the Midstream Assets) or the business operations of the Contributed Entities.

- "Business Day" means any day of the year on which national banking institutions in Dallas, Texas are open to the public for conducting business and are not required or authorized to close.
 - "Capex Budget" means the capital expenditure budget set forth in Exhibit D hereto.
 - "Closing" has the meaning set forth in Section 3.1.
 - "Closing Date" has the meaning set forth in Section 3.1.
 - "Closing Month" means the calendar month in which the Closing occurs.
 - "Closing Statement" has the meaning set forth in Section 2.3(a).
 - "Code" means the Internal Revenue Code of 1986, as amended.
 - "Commercial Agreements" has the meaning set forth in the Company Agreement.
 - "Company" has the meaning set forth in the Preamble.
 - "Company Agreement" has the meaning set forth in the Recitals to this Agreement.
 - "Company Documents" has the meaning set forth in Section 6.2.
 - "Company Indemnified Parties" has the meaning set forth in Section 8.2.
- "Contract" means any written or oral contract, agreement, indenture, note, bond, mortgage, loan, instrument, lease, license, warranty, understanding, arrangement or legally binding commitment or undertaking of any nature.
- "Contributed Entities" means DLK Black River, Longwood RB, Longwood Wolf, Black River Water and Fulcrum Delaware Water Resources, LLC.
 - "Contributed Interests" means 100% of the outstanding equity interests in the Contributed Entities.
- "Contribution Adjustment Amount" means an amount, whether positive or negative, equal to (i) an amount, whether positive or negative, equal to Net Working Capital minus Target Net Working Capital minus (ii) the Operating Amount and calculated in a manner that is consistent with the illustrative calculation in Exhibit C.
- "<u>Current Assets</u>" means (i) cash and cash equivalents, inventory, accounts receivable, prepaid expenses and deposits, including amounts owed to any Contributed Entity associated with any gas imbalances (including fees for penalties) relating to gas deliveries prior to the Closing Date, and (ii) oil, natural gas or natural gas liquid inventories held, as of the Closing Date, in storage tanks associated with the Midstream Assets, but

excluding: (a) the portion of any prepaid expense or deposit of which the Company will not receive the benefit following the Closing Date; (b) receivables owed to any Contributed Entity from its Affiliates unless such receivables are generated by the operations of the Midstream Assets; (c) any prepaid Rights-of-Way costs; and (d) any deferred tax assets, in each case, determined in accordance with GAAP applied using the same accounting methods, practices and principles, with consistent classifications and estimation methods used in the preparation of the applicable Balance Sheet.

"Current Liabilities" means trade accounts payable and accrued current liabilities, including (a) amounts owed by any Contributed Entity associated with any gas imbalances (including fees for penalties) relating to gas deliveries prior to the Closing Date, (b) any unpaid costs and expenses owed by the Contributed Entities as of the Closing Date with respect to the Midstream Assets and (c) accrued and unpaid costs and expenses with respect to the purchase of goods or services relating to the engineering, procurement or construction of the Midstream Assets, but excluding: (w) payables owed by any Contributed Entity to its Affiliates unless such payables relate to the operations of the Midstream Assets; (x) any current obligations related to periods after the Closing Date under leases required to be capitalized in accordance with GAAP; (y) the current portion of any asset retirement obligations of any Contributed Entity and (z) any deferred tax liabilities, in each case, determined in accordance with GAAP applied using the same accounting methods, practices and principles, with consistent classifications and estimation methods used in the preparation of the applicable Balance Sheet.

"Damages" has the meaning set forth in Section 8.2.

"Deductible" has the meaning set forth in Section 8.5(a).

"DLK Black River" means DLK Black River Midstream, LLC, a Texas limited liability company and subsidiary of Longwood.

"Effective Date" means February 1, 2017.

"Environmental Law" means any applicable Law, including the common law, or any Order relating to pollution, the protection, restoration, or remediation of or prevention of harm to the environment (including soil, land, surface or subsurface strata, surface waters, groundwater, drinking water, sediments, ambient air, and plant and animal life) or natural resources, or the protection of human health and safety as such relates to exposure to Hazardous Materials, including Laws and Orders relating to (i) the exposure to, or Releases or threatened Releases of, Hazardous Materials; (ii) the generation, manufacture, processing, distribution, use, treatment, containment, disposal, storage, transport or handling of Hazardous Materials; or (iii) recordkeeping, notification, disclosure and reporting requirements respecting Hazardous Materials.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means, with respect to any Person, any trade or business (whether or not incorporated) which would be considered a single employer with that Person pursuant to Section 414(b), (c), (m) or (o) of the Code and the regulations promulgated under those sections as of any relevant date of determination.

"**Final Allocation**" has the meaning set forth in Section 7.2(h).

"<u>Final Contribution Adjustment Amount</u>" means an amount, whether positive or negative, equal to (i) an amount, whether positive or negative, equal to Final Net Working Capital minus Target Net Working Capital *minus* (ii) the Final Operating Amount and calculated in a manner that is consistent with the illustrative calculation in <u>Exhibit C</u>.

"<u>Final Net Working Capital</u>" means Net Working Capital (i) as shown in Longwood's calculation delivered pursuant to <u>Section 2.3(a)</u> if no notice of disagreement with respect thereto is duly delivered pursuant to <u>Section 2.3(b)</u>; or (ii) if such a notice of disagreement is delivered, (A) as agreed by Five Point and Longwood pursuant to <u>Section 2.3(c)</u> or (B) in the absence of such agreement, as shown in the Independent Accountant's calculation delivered pursuant to <u>Section 2.3(c)</u>.

"Final Operating Amount" means the Operating Amount (i) as shown in Longwood's calculation delivered pursuant to Section 2.3(a) if no notice of disagreement with respect thereto is duly delivered pursuant to Section 2.3(b); or (ii) if such a notice of disagreement is delivered, (A) as agreed by Five Point and Longwood pursuant to Section 2.3(c) or (B) in the absence of such agreement, as shown in the Independent Accountant's calculation delivered pursuant to Section 2.3(c).

"Five Point" has the meaning set forth in the Preamble to this Agreement.

"Five Point Agreements" has the meaning set forth in Section 5.2.

"<u>Fundamental Representations</u>" means the representations and warranties set forth in <u>Sections 4.1, 4.2, 4.4, 4.11(b)</u> (solely with respect to the Commercial Agreements), <u>4.19</u>, <u>5.1</u>, <u>5.2</u>, <u>5.5</u>, <u>5.7</u>, <u>6.1</u>, <u>6.2</u> and <u>6.4</u>.

"GAAP" means generally accepted accounting principles in the United States as of the date hereof, consistently applied.

"Governmental Body" means any (i) nation, state, commonwealth, province, territory or other jurisdiction of any nature or (ii) government or governmental or regulatory body thereof, or political subdivision thereof, whether federal, state, local or foreign, or any agency, instrumentality or authority thereof, or any court or arbitrator (public or private).

"Hazardous Material" means any substance, chemical, material or waste that is (i) defined or regulated as "hazardous," "toxic," "radioactive," a "pollutant," a "contaminant," "petroleum," "oil," or other words of similar meaning and effect under any Law relating to pollution, waste, human health and safety or the environment or (ii) that can form the basis of any Liability under any Law relating to pollution, waste, human health and safety or the environment.

"Hydrocarbons" means oil, natural gas, ethane, propane, ethylene, propylene and other liquid and gaseous hydrocarbons.

"Indebtedness" of any Person means, without duplication, (i) the principal of and premium (if any), and accrued and unpaid interest in respect of indebtedness of such Person for borrowed money, including indebtedness evidenced by notes, debentures, bonds or other similar instruments for the payment of which such Person is responsible or liable; (ii) all obligations of such Person issued or assumed as the deferred purchase price of property (but excluding trade accounts payable incurred in the Ordinary Course of Business and repayable in accordance with customary trade practices and other accrued Current Liabilities); (iii) all obligations of such Person under leases required to be capitalized in accordance with GAAP; (iv) all obligations of such Person for the reimbursement of any obligor on any letter of credit, banker's acceptance or similar credit transaction; and (v) all obligations of the type referred to in clauses (i) through (iv) of any Persons the payment of which such Person is responsible or liable, directly or indirectly, as obligor, guarantor, surety or otherwise, including guarantees of such obligations.

"Indemnified Party" has the meaning set forth in Section 8.4(a).

"Indemnifying Party" has the meaning set forth in Section 8.4(a).

"Independent Accountant" has the meaning set forth in Section 2.3(c).

"Intellectual Property" means all intellectual property rights used by the Company arising from or in respect of the following: (i) all technical information, shop rights, designs, plans, manuals, specifications and other proprietary and nonproprietary technology and data and (ii) all software of the Company.

"Interest" has the meaning set forth in the Company Agreement.

"Interim Expense Amount" means all costs and expenses incurred by the Contributed Entities during the Interim Period (as allocated in accordance with the definition of Operating Amount) in accordance with GAAP (whether accrued on the balance sheet of a Contributed Entity or paid in cash) in connection with the acquisition, ownership, construction, development, operation and maintenance of the Midstream Assets, including, without duplication, (i) capital expenditures and other development costs and expenditures to the extent incurred in the Ordinary Course of Business and in accordance with the CapEx Budget and (ii) Allocable Expenses reasonably incurred during the Interim Period; provided, however, that for the avoidance of doubt, "Interim Expense Amount" shall not include (1) any interest costs, commitment fees (whether for unused borrowings or otherwise), letter of credit fees or other fees relating to Indebtedness or (2) any costs or expenses incurred during the Interim Period to cure any matter that would have constituted a breach or inaccuracy of any representation or warranty of Longwood under Article IV (without regard to any limitation or qualification as to materiality or Material Adverse Effect) as if such representation or warranty has been made as of the date such cost or expenditure is incurred.

"Interim Period" means the period from 12:01 a.m., Dallas, Texas time, on the Effective Date through 11:59 p.m., Dallas, Texas time, on the day prior to Closing Date.

"Interim Revenue Amount" means the sum of all income and revenues of the Contributed Entities in respect of the Midstream Assets earned during the Interim Period (as allocated in accordance with the definition of Operating Amount) in accordance with GAAP (whether accrued on the balance sheet of a Contributed Entity or received in cash).

"IRS" means the United States Internal Revenue Service and, to the extent relevant, the United States Department of Treasury.

"Knowledge of Longwood" means the actual knowledge of those Persons identified on Schedule 1.1(a)(i) without any duty of inquiry.

"Knowledge of Five Point" means the actual knowledge of those Persons identified on Schedule 1.1(a)(ii) without any duty of inquiry.

"Law" means any foreign, federal, state, local or municipal law, statute, constitution, principle of common law, edict, decree, code, ordinance, rule or regulation.

"<u>Legal Proceeding</u>" means any action, suit, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation commenced, brought, conducted or heard by or before any court or other Governmental Body or any arbitrator or arbitration panel.

"<u>Liability</u>" means any debt, loss, damage, adverse claim, liability or obligation (whether direct or indirect, known or unknown, asserted or unasserted, absolute or contingent, accrued or unaccrued, liquidated or unliquidated, or due or to become due, and whether in contract, tort, equity, strict liability or otherwise) and including all costs and expenses relating thereto.

"<u>Lien</u>" means any lien, encumbrance, pledge, mortgage, deed of trust, security interest, claim, lease, charge, option, right of first refusal or other transfer restriction, easement, servitude, proxy, voting trust or similar agreement.

"Limitation Exceptions" has the meaning set forth in Section 8.5(a).

"Longwood" has the meaning set forth in the Preamble to this Agreement.

"Longwood Agreements" has the meaning set forth in Section 4.2.

"Longwood RB" means Longwood RB Pipeline, LLC, a Texas limited liability company and Subsidiary of Longwood.

"Longwood Wolf" means Longwood Wolf Pipeline, LLC, a Texas limited liability company and Subsidiary of Longwood.

"Material Adverse Effect" means any change, effect, event, matter or circumstance that, individually or in the aggregate, has had or would reasonably be expected to have, a material adverse effect on (a) in the case of Longwood, the ability of Longwood to perform its obligations under this Agreement, or (b) in all cases, the business, properties, assets, liabilities, financial condition or results of operations of the Contributed Entities, taken as a whole; provided, however, that for purposes of this clause (b), in no event shall any of the following be taken into account in determining whether there has been or will be a Material Adverse Effect: (i) changes in, or conditions affecting, the economy in the United States or any other country in the world; (ii) changes in, or conditions affecting, the financial markets in the United States or any other country in the world; (iii) acts or failures to act by any Governmental Body; (iv) changes in, or conditions affecting, the oil and gas industry, including changes in the prices of Hydrocarbons; (v) hostilities, acts of war, sabotage, terrorism, military actions or other similar events or any escalation or material worsening of any such hostilities, acts of war, sabotage or terrorism or military actions existing or underway as of the date hereof; (vi) natural disasters, including earthquakes, hurricanes, tornados, wild fires, floods, mud slides, tsunamis, storms and other similar force majeure events; (vii) changes in Laws or GAAP, or in the interpretation thereof; (viii) any failure by any Person to meet any of its financial projections, forecasts, budgets or estimates (it being understood and agreed that the exception in this clause (viii) shall not preclude, prevent or otherwise affect a determination that the facts, circumstances, changes, events, matters, developments, conditions, occurrences or effects giving rise to such failure should be deemed to constitute, or be taken into account in determining whether there has been, a Material Adverse Effect); (ix) the execution of this Agreement or the announcement, disclosure or pendency of the transactions contemplated by this Agreement (including the impact thereof on relationships, contractual or otherwise, with customers, suppliers, licensors, distributors, vendors, partners or employees); provided, that the exception in this clause (ix) shall not apply to references to "Material Adverse Effect" in the representations and warranties set forth in Section 4.3; and (x) compliance with the terms of, or the taking of any action required by, any Transaction Agreement, as applicable, or taking any action at the written request of Five Point; provided, that the exception in this clause (x) shall not apply to references to "Material Adverse Effect" in the representations and warranties set forth in Section 4.3, in each case of clauses (i) through (vii), to the extent that such acts or events do not disproportionately affect the Contributed Entities, taken as a whole, relative to other similarly situated participants in the industries of the Contributed Entities.

"Material Contracts" has the meaning set forth in Section 4.11(a).

"Midstream Assets" means the Wolf Assets and the Rustler Breaks Assets.

"Net Working Capital" means an amount, whether positive or negative, equal to the Current Assets of the Contributed Entities, as a whole, minus the Current Liabilities of the Contributed Entities, as a whole, in each case as of the close of business on the day prior to the Closing Date and calculated in a manner that is consistent with the illustrative calculation in Exhibit C; provided, that, notwithstanding anything herein to the contrary, for purposes of calculating Net Working Capital for a Closing Date that occurs on a day other than the first day of a month, for each component of Net Working Capital that

directly corresponds to, and is derived from, a component of the Operating Amount, the pro rata amount used for such component in the calculation of Operating Amount shall be the corresponding amount used for purposes of calculating Net Working Capital.

"NGA" has the meaning set forth in Section 4.18.

"Non-Party Affiliates" has the meaning set forth in Section 9.8.

"Operating Amount" means an amount, whether positive or negative, equal to the Interim Revenue Amount minus the Interim Expense Amount and calculated in a manner that is consistent with the illustrative calculation in Exhibit C; provided, that, notwithstanding anything herein to the contrary, for purposes of computing the Operating Amount for a Closing Date that occurs on a day other than the first day of a month, the components of the Operating Amount related to the operations of the Company during the Closing Month will be allocated to the Interim Period on a pro rata basis using a fraction, the numerator of which is the number of days in the Closing Month prior to the Closing Date and the denominator of which is the number of days in the Closing Month.

"Order" means any order, injunction, judgment, decree, ruling, writ, assessment or arbitration award of a Governmental Body.

"Ordinary Course of Business" means the ordinary and usual course of normal day-to-day operations of the Contributed Entities through the date hereof, consistent with past practices of the Contributed Entities and consistent with the customary industry practices of entities who own assets similar to the Midstream Assets.

"Organizational Documents" means, with respect to any entity, the documents governing the formation, operation and governance of such entity, including (i) in the case of a corporation, the articles or certificate of incorporation and bylaws of such corporation, (ii) in the case of a limited partnership, the certificate of formation and the limited partnership agreement of such limited partnership and (iii) in the case of a limited liability company, the certificate of formation and limited liability company agreement of such limited liability company.

"Owned Property" and "Owned Properties" have the meanings set forth in Section 4.8(a).

"Party" means each of Longwood, the Company and Five Point, individually, and "Parties" means Longwood, the Company and Five Point, collectively.

"<u>Performance Incentive Agreement</u>" means that certain letter agreement to be entered into as of the Closing Date between the Company, Five Point, FP MMP Partners LLC, a Delaware limited liability company, Longwood and MRC Permian Company, a Texas corporation.

"Permits" means the approvals, authorizations, consents, licenses, permits or certificates of a Governmental Body.

"Permitted Liens" means (i) all defects, exceptions, restrictions, easements, rights of way and encumbrances disclosed in policies of title insurance which have been made available to Five Point, (ii) Liens for Taxes, assessments or other governmental charges not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings; provided, that an appropriate reserve is established therefor, (iii) mechanics', carriers', workers', repairers' and similar Liens arising or incurred in the Ordinary Course of Business that are not material to the business, operations and financial condition of the property so encumbered and that secure obligations that are not delinquent, (iv) zoning, entitlement and other land use and environmental regulations by any Governmental Body; provided, that such regulations have not been violated, (v) title of a lessor under a capital or operating lease; provided, that such lease has been complied with, (vi) encumbrances on title arising by operation of any applicable United States federal, state or foreign securities Laws; provided, that such Laws have not been violated and (vii) such other imperfections in title, charges, easements, restrictions and encumbrances which do not materially detract from the value of or in any manner materially impair or interfere with the present use of any asset subject thereto or affected thereby.

"<u>Person</u>" means any individual, corporation, partnership, limited liability company, limited partnership, firm, joint venture, association, joint-stock company, trust, unincorporated organization, Governmental Body or other entity.

"Proposed Allocation" has the meaning set forth in Section 7.2(h).

"Purchasers" has the meaning set forth in the Preamble.

"Purchaser Indemnified Parties" has the meaning set forth in Section 8.3.

"Real Property Interests" means the Real Property Leases, Rights-of-Way and all other surface use and sub-surface use agreements, licenses and leases and real property interests (other than the Owned Properties) related to the Midstream Assets.

"Real Property Lease" and "Real Property Leases" have the meanings set forth in Section 4.8(a).

"Release" means any release, spill, emission, discharge, leaking, pouring, dumping or emptying, pumping, injection, deposit, disposal, dispersal, leaching or migration into the indoor or outdoor environment (including, without limitation, soil, ambient air, surface water, groundwater and surface or subsurface strata) or into or out of any property, including the movement of Hazardous Materials through or in the air, soil, surface water, groundwater or property.

"Retained Agreements" has the meaning set forth in Section 7.6.

"Rights-of-Way" has the meaning set forth in Section 4.8(a).

"Rustler Breaks Assets" means (i) the Rustler Breaks Plant, (ii) the certain gas gathering and compression system and produced water gathering assets located in Eddy

County, New Mexico, as set forth in more detail on <u>Exhibit A-1</u> hereto and (iii) those certain produced water disposal assets located in Eddy County, New Mexico, as set forth in more detail on <u>Exhibit A-1</u> hereto.

"Rustler Breaks Plant" means that certain gas processing plant located in Eddy County, New Mexico, as more fully described on Exhibit A-2 hereto.

"Securities" means, with respect to a Purchaser, the Interest of such Purchaser in the Company following the Closing as set forth in the Company Agreement.

"Securities Act" means the Securities Act of 1933, as amended.

"Specified Matters" has the meaning set forth in Section 8.2.

"Straddle Period" has the meaning set forth in Section 7.2(b).

"Subsidiary" means, with respect to any Person, any corporation, partnership or other entity (i) of which equity securities or other ownership interests having ordinary voting power to elect a majority of the board of directors or other similar managing body of such corporation, partnership or other entity or having the right to receive more than 50% of the distributions to be made by such corporation, partnership or other entity (either generally or upon liquidation of such corporation, partnership or other entity) are at the time owned by such Person or (ii) the management of which is otherwise controlled by such Person.

"Target Net Working Capital" means an amount equal to \$0.00.

"<u>Tax</u>" or "<u>Taxes</u>" means all taxes, levies, imposts and similar assessments paid or payable to a Governmental Body, including, but not limited to, all net income, gross income, gross receipts, sales, use, value added, services, occupation, transfer, franchise, capital stock, profits, license, withholding, payroll, employment, unemployment, excise, estimated, severance, stamp or occupancy taxes, governmental fee or other like assessment or charge of any kind whatsoever, together with any interest, penalty or addition to any such tax or other charge.

"<u>Tax Return</u>" means any return, report or statement filed or required to be filed with respect to any Tax (including any attachments thereto, and any amendment thereof), including any information return, claim for refund, amended return or declaration of estimated Tax.

"Taxing Authority" means the IRS and any other Governmental Body responsible for the administration of any Tax.

"Third Party Claim" has the meaning set forth in Section 8.4(a).

"<u>Transaction Agreements</u>" means this Agreement, the Company Agreement, the Performance Incentive Agreement and the Commercial Agreements.

"Treasury Regulations" means the regulations promulgated under the Code, as such regulations may be amended from time to time.

"<u>Wolf Assets</u>" means (i) that certain natural gas, crude oil and water gathering system and related assets located in Loving County, Texas, as set forth in more detail on <u>Exhibit A-3</u> hereto and (ii) those certain produced water disposal assets located in Loving County, Texas, as set forth in more detail on <u>Exhibit A-3</u> hereto.

1.2 Other Definitional and Interpretive Matters

(a) Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:

<u>Breach</u>. When the term "breach" is used in this Agreement in the context of a covenant or agreement of a Person, the term includes a breach or failure to perform.

<u>Calculation of Time Period</u>. When calculating the period of time before which, within which or following which, any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.

<u>Dollars</u>. Any reference in this Agreement to \$ or "dollars" shall mean U.S. dollars.

Exhibits/Schedules. The Exhibits and Schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement. Any matter or item disclosed on one Schedule shall be deemed to be disclosed on each other Schedule of which it is reasonably apparent from the face of such disclosure that such disclosure is applicable to another Schedule. Disclosure of any item on any Schedule shall not constitute an admission or indication that such item or matter is material or would have a Material Adverse Effect. No disclosure on a Schedule relating to a possible breach or violation of any Contract, Law or Order shall be construed as an admission or indication that a breach or violation exists or has actually occurred. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.

<u>Gender and Number</u>. Any reference in this Agreement to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.

<u>Headings</u>. The provision of a Table of Contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "<u>Section</u>" are to the corresponding Section of this Agreement unless otherwise specified.

<u>Herein</u>. The words such as "<u>herein</u>," "<u>hereinafter</u>," "<u>hereof</u>," and "<u>hereunder</u>" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.

<u>Including</u>. The word "<u>including</u>" or any variation thereof means (unless the context of its usage otherwise requires) "<u>including</u>, <u>without</u> <u>limitation</u>" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

(b) The Parties have participated jointly in the negotiation and drafting of this Agreement and, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement.

ARTICLE II PURCHASE AND SALE OF THE SECURITIES

- 2.1 <u>Purchase and Sale of the Securities</u>. In accordance with and subject to the terms and conditions set forth herein, at the Closing, the Company shall issue and sell to the Purchasers, and the Purchasers shall severally purchase from the Company, the Securities free and clear of any and all Liens (other than restrictions on transfer under the Company Agreement or under state or federal securities laws).
- 2.2 <u>Consideration</u>. As consideration for the Securities, at the Closing the Purchasers shall contribute, pay and commit to pay certain assets and cash, as applicable, to the Company as follows:
 - (a) At the Closing, Longwood shall (i) contribute to the Company the Contributed Interests (which Contributed Interests have an agreed initial Book Basis of \$350,000,000) and cash in the amount of \$5,100,000, which contributions, when made, shall be treated as Capital Contributions (as defined in the Company Agreement) to the Company pursuant to the Company Agreement, and (ii) commit to pay to the Company pursuant to Section 4.1 of the Company Agreement up to \$71,400,000 in cash; and
 - (b) At the Closing, Five Point shall (i) contribute to the Company cash in the amount of \$176,400,000, which contribution, when made, shall be treated as a Capital Contribution to the Company pursuant to the Company Agreement, and (ii) commit to pay to the Company pursuant to Section 4.1 of the Company Agreement up to \$68,600,000 in cash.

2.3 Contribution Adjustment Amount.

(a) Within 120 days after the Closing Date, Longwood shall cause to be prepared and delivered to the Company and Five Point a statement (the "Closing Statement") setting forth Longwood's calculation of (i) Net Working Capital, (ii) the Operating Amount and (iii) the Contribution Adjustment Amount.

- (b) If Five Point disagrees with Longwood's calculation of the Contribution Adjustment Amount (or any component thereof) delivered pursuant to Section 2.3(a), Five Point may, within 60 days after delivery of the Closing Statement, deliver a notice to Longwood stating that Five Point disagrees with any such calculation and specifying in reasonable detail those items or amounts as to which Five Point disagrees and the basis therefor. If Five Point fails to deliver a notice by the conclusion of such 60-day period, Five Point shall be deemed to have agreed with all items and amounts contained in the Closing Statement and the calculation of the Contribution Adjustment Amount delivered pursuant to Section 2.3(a) shall be the Final Contribution Adjustment Amount.
- (c) If a notice of disagreement shall be timely delivered pursuant to Section 2.3(b), Five Point and Longwood shall, during the 60 days following such delivery, use their commercially reasonable efforts to reach agreement on the disputed items or amounts in order to determine, as may be required, the amount of Net Working Capital, the Operating Amount or the Contribution Adjustment Amount, as applicable. If during such period Five Point and Longwood are unable to reach such agreement, they shall promptly thereafter cause Pricewaterhouse Coopers LLP (or if unable or unwilling to accept its mandate, an independent accounting firm to be mutually agreed upon by Five Point and Longwood) (Pricewaterhouse Coopers LLP or such other independent accounting firm, as the case may be, the "Independent Accountant") to review this Agreement and the disputed items or amounts for the purpose of calculating the Contribution Adjustment Amount (it being understood that in making such calculation, the Independent Accountant shall be functioning as an expert and not as an arbitrator). Each of Five Point and Longwood agree that it shall not engage, or agree to engage the Independent Accountant to perform any services other than as the Independent Accountant pursuant hereto until the Closing Statement and the Contribution Adjustment Amount have been finally determined pursuant to this Section 2.3. Each of Five Point and Longwood agrees to execute, if requested by the Independent Accountant, a reasonable engagement letter. Five Point and Longwood shall cooperate with the Independent Accountant and promptly provide all documents and information requested by the Independent Accountant. In making such calculation, the Independent Accountant shall consider only those items or amounts in the Closing Statement and the calculation of the Contribution Adjustment Amount as to which Five Point has disagreed in its notice of disagreement duly delivered pursuant to Section 2.3(b). The Independent Accountant shall deliver to Five Point and Longwood, as promptly as practicable (but in any case no later than 30 days from the date of engagement of the Independent Accountant), a report setting forth such calculation. Such report shall be final and binding upon Five Point and Longwood (absent fraud or manifest error), and neither Five Point nor Longwood shall seek further recourse to courts or other tribunals, other than to enforce such report. Judgment may be entered to enforce such report in any court of competent jurisdiction. Any fees and expenses of the Independent Accountant shall be borne one-half by Five Point and one-half by Longwood.
- (d) Longwood shall, and shall cause its representatives to, cooperate with Five Point with respect to any review of the Closing Statement prior to the objection deadline referred to in <u>Section 2.3(b)</u>, including the making available, to the extent reasonably necessary, of books, records, work papers and personnel during regular business hours upon reasonable advance notice and under reasonable circumstances, subject to restrictions under applicable Law.

(e) (i) If the Final Contribution Adjustment Amount, as finally determined pursuant to this <u>Section 2.3</u>, is a positive amount (i.e., greater than \$0.00), the Company shall pay to Longwood, out of the Initial Capital Contributions (as defined in the Company Agreement), cash in an amount equal to the Final Contribution Adjustment Amount, or (ii) if the Final Contribution Adjustment Amount is a negative amount (i.e., less than \$0.00), as finally determined pursuant to this <u>Section 2.3</u>, Longwood shall pay to the Company cash in an amount equal to the Final Contribution Adjustment Amount (expressed as a positive amount).

ARTICLE III CLOSING; CLOSING DELIVERABLES

3.1 <u>Closing</u>. The closing (the "<u>Closing</u>") of the purchase and sale of the Securities and the other transactions contemplated by this Agreement shall be held at 9:00 a.m., Dallas, Texas time, on the date hereof (the "<u>Closing Date</u>"), but effective as of 12:01 a.m., Dallas, Texas time, at such location or locations as the parties shall mutually agree.

3.2 Closing Deliverables.

- (a) At the Closing, Longwood shall deliver (or cause its applicable Affiliate to deliver, in the case of the Performance Incentive Agreement and the Commercial Agreements) each of the following:
 - (i) to the Company, cash in the amount of \$5,100,000;
 - (ii) to the Company, a duly executed counterpart of an Assignment and Assumption Agreement substantially in the form of <u>Exhibit B</u> (the "Assignment and Assumption Agreement");
 - (iii) to the Company, duly executed counterparts of each Commercial Agreement;
 - (iv) to the Company, a certificate, duly executed and acknowledged by Longwood (or, in the event Longwood is disregarded as an entity separate from its owner for U.S. federal income tax purposes, by its owner) in the form prescribed by Treasury Regulations Section 1.1445-2(b)(2)(iv) stating that Longwood (or, in the event Longwood is disregarded as an entity separate from its owner for U.S. federal income tax purposes, its owner) is not a "foreign person" within the meaning of Section 1445 of the Code;
 - (v) to Five Point, duly executed counterparts of the Company Agreement and the Performance Incentive Agreement; and
 - (vi) to Five Point, such other documents as Five Point shall reasonably request in connection with the consummation of the transactions contemplated by <u>Article II</u>.

- (b) At the Closing, Five Point shall deliver each of the following:
 - (i) to the Company, cash in the amount of \$176,400,000;
 - (ii) to the Company, a certificate, duly executed and acknowledged by Five Point (or, in the event Five Point is disregarded as an entity separate from its owner for U.S. federal income tax purposes, by its owner) in the form prescribed by Treasury Regulations Section 1.1445-2(b)(2)(iv) stating that Five Point (or, in the event Five Point is disregarded as an entity separate from its owner for U.S. federal income tax purposes, its owner) is not a "foreign person" within the meaning of Section 1445 of the Code;
 - (iii) to Longwood, duly executed counterparts of the Company Agreement and the Performance Incentive Agreement; and
 - (iv) to Longwood, such other documents as Longwood shall reasonably request in connection with the consummation of the transactions contemplated by Article II.
- (c) At the Closing, the Company shall deliver each of the following:
 - (i) to Longwood, duly executed counterparts of the Assignment and Assumption Agreement and each Commercial Agreement;
 - (ii) to Five Point and to Longwood, duly executed counterparts of the Performance Incentive Agreement; and
 - (iii) to the Purchasers, such other documents, certificates, filings or other agreements as either Purchaser shall reasonably request in connection with the consummation of the transactions contemplated by <u>Article II</u>.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF LONGWOOD

Longwood hereby represents and warrants to the Company and Five Point that:

4.1 <u>Organization and Good Standing</u>. Each of Longwood and each Contributed Entity is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Texas, and Longwood and each Contributed Entity has all requisite limited liability company power and authority to own, lease and operate its properties and to carry on its business as now conducted. Except as set forth on <u>Schedule 4.1</u>, the Contributed Entities are duly qualified or authorized to do business and are in good standing under the laws of each jurisdiction in which they own or lease real property and each other jurisdiction in which the conduct of their business or the ownership of their properties requires such qualification or authorization, except where the failure to be so qualified, authorized or in good standing would not, individually or in the aggregate, have a Material Adverse Effect.

4.2 <u>Authorization of Agreement</u>. Longwood has all requisite power and authority to execute and deliver this Agreement and each other Transaction Agreement to which it is a party, including the Company Agreement (the "<u>Longwood Agreements</u>"), and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Longwood Agreements and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all requisite limited liability company action on the part of Longwood. This Agreement has been, and each of the other Longwood Agreements will be at or prior to the Closing, duly and validly executed and delivered by Longwood and (assuming the due authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes, and each Longwood Agreement, when so executed and delivered will constitute, the legal, valid and binding obligations of Longwood, enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity).

4.3 Conflicts; Consents of Third Parties.

- (a) None of the execution and delivery by Longwood of this Agreement or the Longwood Agreements, the consummation of the transactions contemplated hereby or thereby, or compliance by Longwood with any of the provisions hereof or thereof will conflict with, or result in any violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation or acceleration of any obligation or to a loss of a benefit under, or give rise to any obligation of Longwood to make any payment under, or to the increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or result in the creation of any Liens, other than Permitted Liens, upon any of the material assets of Longwood or any of the Contributed Entities under, any provision of (i) the Organizational Documents of Longwood or any of the Contributed Entities are a party or by which any of the properties or assets of Longwood or any of the Contributed Entities is bound; (iii) any Order of any Governmental Body applicable to Longwood or any of the Contributed Entities or by which any of the properties or assets of any of the Contributed Entities are bound; or (iv) any applicable Law, other than, in the case of clauses (ii), (iii) and (iv), such conflicts, violations, defaults, terminations or cancellations, that would not, individually or in the aggregate, have a Material Adverse Effect.
- (b) Except as set forth on Schedule 4.3(b), no consent, waiver, approval, Order, Permit or authorization of, or declaration or filing with, or notification to, any Person or Governmental Body is required on the part of Longwood in connection with the execution and delivery of this Agreement or the other Longwood Agreements or the compliance by Longwood with any of the provisions hereof or thereof, or the consummation of the transactions contemplated hereby or thereby or the taking by Longwood of any other action contemplated hereby, except for the consents, waivers,

approvals, Orders, Permits, authorizations of, or declarations or filings with any Governmental Body which (i) have been obtained or (ii) are customarily made or obtained following the Closing.

4.4 Title to the Contributed Interests; Capitalization.

- (a) Longwood is the record and beneficial owner of the Contributed Interests, free and clear of any and all Liens (other than restrictions on transfer that may be imposed under state or federal securities laws or the Organizational Documents of any Contributed Entity). The transfer, assignment and delivery of the Contributed Interests to the Company will convey to the Company good and marketable title to the Contributed Interests, free and clear of any and all Liens (other than restrictions on transfer that may be imposed under state or federal securities laws or the Organizational Documents of any Contributed Entity).
- (b) All Contributed Interests are duly authorized, validly issued, fully paid and non-assessable, and are not subject to and were not issued in violation of any preemptive rights or any other agreement, arrangement or commitment, including a purchase option, call option, right of first refusal, subscription right or any similar right, and were issued in compliance with applicable federal and state securities Laws. Except as set forth in the Organizational Documents of each Contributed Entity, there is no outstanding purchase option, call option, right of first refusal, subscription right, phantom equity, equity appreciation or any similar right with respect to the Contributed Interests or any other equity interest in any of the Contributed Entities. Except for the Organizational Documents of each Contributed Entity, there are no agreements, arrangements, commitments or claims relating to the Contributed Interests or obligating a Contributed Entity or any other Person to issue, transfer or sell or cause to be issued, transferred or sold any Contributed Interest or obligating a Contributed Entity or any such Person to grant, extend or enter into any such option, warrant, call, preemptive right, phantom equity or equity appreciation, subscription, commitment or claim and no such instruments are outstanding. Except for the Contributed Interests and as set forth in the Organizational Documents of each Contributed Entity, no voting or non-voting securities or any other equity interests are issued, reserved for issuance or outstanding with respect to a Contributed Entity, and there are no rights or instruments outstanding to acquire any such securities or equity interests in any Contributed Entity. There are no convertible securities of any Contributed Entity outstanding which upon conversion would require the issuance of any equity interests of any Contributed Entity or other securities convertible into shares of equity interests of any Contributed Entity. Except as set forth in the Organizational Documents of each Contributed Entity, there is no obligation (contingent or otherwise) of any Contributed Entity to purchase, redeem or otherwise acquire any of the Contributed Interests or other securities or any interest therein or to pay any dividend or make any other distribution in respect thereof.
- (c) Schedule 4.4(c) sets forth a list of each Subsidiary of Longwood immediately prior to the consummation of the transactions contemplated by this Agreement. Except as set forth on Schedule 4.4(c), no Contributed Entity owns any stock, membership interest, partnership interest or other equity interest in any Person.

4.5 Financial Statements and Related Information.

- (a) Longwood has made available to Five Point copies of the unaudited consolidated balance sheets of each Contributed Entity at December 31, 2016, in each case, which sets forth in reasonably sufficient detail such Contributed Entity's assets and liabilities (such balance sheets, including any related notes and schedules thereto, are referred to herein as the "Balance Sheets"). Except as set forth in the notes thereto and as disclosed on Schedule 4.5(a), the Balance Sheets have been prepared in accordance with GAAP (subject, in each case, to any normal year-end adjustments and the absence of footnote disclosures) and present fairly in all material respects the financial position of the Contributed Entities at December 31, 2016.
- (b) Except as set forth in <u>Schedule 4.5(b)</u>, no Contributed Entity has any Indebtedness, obligations or Liabilities of any kind other than those (i) fully reflected in, reserved against or otherwise described in the Balance Sheets or the notes thereto or (ii) immaterial to such Contributed Entity and incurred in the Ordinary Course of Business since the date of the respective Balance Sheet.
- 4.6 <u>Absence of Certain Developments</u>. Except as contemplated by this Agreement or set forth on <u>Schedule 4.6</u>, (a) since the date of the Balance Sheets, the Contributed Entities have conducted their business only in the Ordinary Course of Business, and (b) since the date of the Balance Sheet, there has not occurred any event, change, condition, occurrence or circumstance that has had, or could reasonably be expected to have, a Material Adverse Effect on any Contributed Entity.

4.7 <u>Taxes</u>. Except as set forth on <u>Schedule 4.7</u>:

- (a) As of the Closing Date, the Contributed Entities and any predecessor thereof (i) are (and at all times since their formation have been), for U.S. federal and applicable state or local income Tax purposes (excluding Texas franchise tax purposes), partnerships or entities that are treated as disregarded pursuant to Treasury Regulation Section 301.7701-2 and (ii) have not made any filings with any Taxing Authority, including Form 8832 with the IRS, to be treated as an association taxable as a corporation for income Tax purposes.
- (b) The Contributed Entities have timely filed (or have had timely filed on their behalf) with the appropriate Taxing Authorities all Tax Returns required to have been filed by them (taking into account for this purpose any extensions), and such Tax Returns were and are true, complete and correct in all material respects. All Taxes shown to be due on the Tax Returns referred to in this Section 4.7(b), and any and all Taxes otherwise due and payable (regardless of whether shown on any Tax Return) by the Contributed Entities have been timely paid in full (taking into account for this purpose any extensions).
- (c) No claim has been made by any Taxing Authority in a jurisdiction where the Contributed Entities do not file Tax Returns or pay Taxes that any Contributed Entity is or may be required to file Tax Returns or pay Taxes, respectively, in that jurisdiction.

- (d) As of the Closing, there are (i) no pending or, to the Knowledge of Longwood, threatened proposed material adjustments, deficiencies, underpayment or assessment of Taxes from any Taxing Authority with respect to any Contributed Entity, (ii) no ongoing audits or examinations or, to the Knowledge of Longwood, threatened proposed audits or examinations of any Tax Returns or Tax liability of any Contributed Entity and (iii) no Tax liens on the assets of any Contributed Entity other than Liens for Taxes not yet due and payable.
- (e) As of the Closing, (i) no Contributed Entity is the subject of (or a party to) any agreement or arrangement with a Taxing Authority requiring Longwood, any Contributed Entity or any of their Affiliates to employ a certain number of persons or take (or refrain from taking) any other action in order to secure Tax benefits not otherwise available, and (ii) no Contributed Entity is a party to a Tax indemnity, sharing or allocation agreement (other than the agreement as set forth in Section 9.6 of the Company Agreement) with any other Person that might require any Contributed Entity to a make a payment to another Person with respect to Taxes; *provided, however*, that an agreement shall not be considered to be described in this Section 4.7(e) unless such agreement principally relates to Taxes.
- (f) No Contributed Entity will be required to include or accelerate the recognition of any item in income, or exclude or defer any deduction or other tax benefit, in each case in any taxable period (or portion thereof) ending after the Closing Date, as a result of any change in method of accounting, closing agreement, intercompany transaction, installment sale or the receipt of any prepaid amount, in each case prior to Closing.
- (g) Other than any Texas franchise Taxes of a "combined group" that includes Matador Resources Company, MRC Energy Company, Longwood and each Contributed Entity, each Contributed Entity has no liability for the Taxes of another Person (i) as a result of being or having been a member of an affiliated, consolidated, combined, unitary or other similar group prior to the Closing Date or (ii) as a transferee or successor based on an event or transaction that occurred at or prior to the Closing or as a result of the Closing.
- (h) Longwood (or, in the event Longwood is disregarded as an entity separate from its owner for U.S. federal income tax purposes, its owner) is not a "foreign person" as defined in Section 1445 of the Code.
- (i) No Contributed Entity has requested or granted any requests, or entered into any agreements, or provided any consents or waivers to extend the statutory period of limitations applicable to the assessment of any Taxes.
- (j) For the avoidance of doubt, the representations and warranties made in this <u>Section 4.7</u> are the only representations and warranties made by Longwood with

respect to matters related to Taxes. Nothing in this <u>Article IV</u> or otherwise in this Agreement shall be construed as a representation or warranty regarding, or a guarantee of, and cannot be relied upon with respect to, (i) the amount or availability of any credit, loss or other Tax attribute of Longwood or any Contributed Entity after the Closing (except for the representations and warranties included in <u>Section 4.7(a)</u>), (ii) the Taxes of the Company or any of its Subsidiaries (including the Contributed Entities) attributable to any Tax period (or a portion thereof) beginning after the Closing Date (except for the representations and warranties included in <u>Section 4.7(a)</u>, (e), (f), and (g)), or (iii) whether an action taken or position adopted by Longwood, any Contributed Entity or their Affiliates prior to Closing with respect to a taxable period (or portion thereof) ending on or prior to the Closing Date is permitted to be taken or adopted by Five Point, the Company or their Affiliates after Closing with respect to a Tax period (or portion thereof) beginning after the Closing Date (and Five Point and the Company shall make their own determinations in this regard).

4.8 Real Property.

- (a) <u>Schedule 4.8(a)</u> sets forth a complete list of (i) all material real property and material interests in real property owned, directly or indirectly, in fee by each Contributed Entity (individually, an "<u>Owned Property</u>" and collectively, the "<u>Owned Properties</u>"), (ii) all material rights-of-way, easements, licenses and servitudes in favor of any Contributed Entity (the "<u>Rights-of-Way</u>") and (iii) all leases of real property by any Contributed Entity involving annual payments of \$5,000 or greater as lessee or lessor (individually, a "<u>Real Property Lease</u>" and collectively, the "<u>Real Property Leases</u>").
- (b) Except as set forth on Schedule 4.8(b) and Permitted Liens, each Contributed Entity has marketable title to its respective Owned Property, Rights-of-Way and Real Property Leases, free and clear of Liens.
- (c) Except as set forth on Schedule 4.8(c) and Permitted Liens, no Contributed Entity has received any written notice of any claim, default or event that with notice or lapse of time, or both, would constitute a default by any Contributed Entity under any of the Real Property Interests held by it that would result in a termination of any Real Property Interests or that would result in such Contributed Entity not having marketable title in connection with any Real Property Interests.

4.9 Midstream Assets.

- (a) Except for Permitted Liens, and except as set forth on Schedule 4.9(a), each Contributed Entity has marketable title, or a valid leasehold interest, to all of the material items of tangible personal property used by it in the operation of the Midstream Assets, free and clear of any and all Liens. Schedule 4.9(a) lists all Midstream Assets that are owned or leased by Longwood or any of its Affiliates (other than the Contributed Entities).
- (b) Except as set forth on <u>Schedule 4.9(b)</u>, the Midstream Assets have been owned, constructed, maintained and, since the applicable in-service date in respect

thereof, operated in a good and workmanlike manner and in all material respects in accordance with customary practices in the oil and gas and salt water disposal industries and all applicable Laws.

- (c) Except as set forth on Schedule 4.9(c), the Midstream Assets constitute all of the material assets and properties that are owned by or held, as of the date of this Agreement, by Longwood or any Affiliate of Longwood. To the Knowledge of Longwood, except as set forth on Schedule 4.9(c), the Midstream Assets constitute all of the material assets sufficient to permit the Company to perform its obligations under the Commercial Agreements and to otherwise own and operate the assets described on Exhibits A-1, A-2 and A-3 in the Ordinary Course of Business.
- 4.10 <u>Intellectual Property</u>. Except as set forth on <u>Schedule 4.10</u>, each Contributed Entity owns or has valid licenses to use all Intellectual Property used by it in the Ordinary Course of Business, except to the extent the failure to be the owner or the valid licensee would not, individually or in the aggregate, have a Material Adverse Effect. Except as set forth on <u>Schedule 4.10</u>, to the Knowledge of Longwood, (i) the material Intellectual Property used by any Contributed Entity is not the subject of any challenge received by Longwood or any Contributed Entity in writing and (ii) neither Longwood nor any Contributed Entity has received any written notice of any default or any event that with notice or lapse of time, or both, would constitute a default under any material Intellectual Property license to which any Contributed Entity is a party or by which it is bound.

4.11 Material Contracts.

- (a) Other than this Agreement and the Company Agreement, <u>Schedule 4.11(a)</u> sets forth all of the following Contracts to which any Contributed Entity is, or as of the Closing will be, a party or by which any of the Midstream Assets are, or as of the Closing will be, bound (collectively, the "<u>Material Contracts</u>"):
 - (i) the Commercial Agreements;
 - (ii) any Real Property Lease;
 - (iii) any pipeline interconnection agreements;
 - (iv) any lease of personal property that requires or is reasonably expected to require, in accordance with its terms, payments from any Contributed Entity in excess of \$50,000 in any twelve-month period;
 - (v) any Contract that requires or is reasonably expected to require, in accordance with its terms, payments to or from any Contributed Entity in excess of \$50,000 in any twelve-month period other than any Contract that is otherwise disclosed on <u>Schedule 4.11(a)</u>;
 - (vi) Contracts with any Affiliates of any Contributed Entity (including Longwood) or any current or former officer or director of Longwood, such Contributed Entity or their Affiliates, on the one hand, and such Contributed Entity, on the other hand;

- (vii) Contracts that restrict the ability of any Contributed Entity to compete with any other Person or engage in any line of business or which materially limit or restrict the right or ability to operate the Midstream Assets;
- (viii) Contracts for (A) the purchase or sale of any material assets, including purchase orders for material equipment, or (B) the sale of equity interests in any Contributed Entity (other than this Agreement) or the merger, consolidation or reorganization of any Contributed Entity;
- (ix) Contracts relating to any acquisition (after the date hereof) by any Contributed Entity of any operating business or the capital stock of any other Person, including any associated confidentiality agreements;
- (x) Contracts relating to the incurrence, assumption or guarantee of Indebtedness, the making of any advances or loans or imposing a Lien on any of the Midstream Assets;
- (xi) Contracts relating to any partnership, strategic alliance or joint venture or any sharing of revenues, profits, losses, costs or liabilities; and
- (xii) Contracts that include any obligation of the Company or any Contributed Entity to make payments, contingent or otherwise, arising out of the prior acquisition or disposition of any asset or business.
- (b) Longwood has made available to Five Point correct and complete copies (or, in the case of the Commercial Agreements, execution versions) of (A) all written Material Contracts and (B) all Organizational Documents of each Contributed Entity, including, in each case, all amendments thereto. Except as set forth on Schedule 4.11(b)(i), there are no oral Contracts binding on any Contributed Entity. Each of the Material Contracts is, or as of the Closing will be, in full force and effect and is, or as of the Closing will be, the legal, valid and binding obligation of the applicable Contributed Entity or the applicable Affiliate of Longwood, as applicable, and, to the Knowledge of Longwood, each other party thereto, enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity). No Contributed Entity is in breach of any Material Contract in any material respect, nor, to the Knowledge of Longwood, is any other party to any such Material Contract in breach thereof. Except as set forth on Schedule 4.11(b)(ii), no event has occurred, which after notice or lapse of time, or both, would constitute a material breach or other default by any Contributed Entity under any Material Contract, or to the Knowledge of Longwood, any other party to any such Material Contract.

4.12 Labor; Employee Matters; Employee Benefit Plans.

(a) There are no, nor have there ever been, individuals employed by any Contributed Entity. To the Knowledge of Longwood, (i) no Contributed Entity or any of its Affiliates has misclassified any Person as an independent contractor in respect of such

services to any Contributed Entity rather than as an employee under any Laws, (ii) there are no pending or threatened Legal Proceedings with respect to any such misclassification and (iii) no facts or circumstances exist that are reasonably expected to give rise to any such Legal Proceeding.

- (b) There are no material complaints, charges or claims against Longwood or any of its Affiliates pending or, to the Knowledge of Longwood, threatened that could be brought or filed with any Governmental Body based on, arising out of, in connection with or otherwise relating to the employment or termination of employment or failure to employ by Longwood or any of its Affiliates, of any individual provider of services to a Contributed Entity. There are no existing, or to the Knowledge of Longwood, threatened employment or labor disputes subject to any grievance procedure, arbitration, litigation or other Legal Proceeding, with respect to any individual provider of services to a Contributed Entity.
- (c) Neither Longwood nor any of its Affiliates is, or has within the five years prior to the date hereof, been a party to any collective bargaining agreement, works council agreement or other labor union contract, and, to the Knowledge of Longwood, there has not been any threatened union organization, decertification or other labor activities with respect to any individual provider of services to Longwood or its Affiliates during such period, and, to the Knowledge of Longwood, no event has occurred or circumstance exists that may provide the basis for such activity.
- (d) To the Knowledge of Longwood, no employee of, or individual provider of services to, Longwood or any of its Affiliates is in violation in any material respect of any term of any nondisclosure agreement, common law nondisclosure obligation, non-competition agreement, restrictive covenant or other similar obligation, in each case: (i) to Longwood or any Contributed Entity or (ii) to a former employer of any such employee or individual provider of services relating (A) to the right of any such employee to be employed by, or individual service provider to provide services to, Longwood or any Contributed Entity or (B) to the knowledge or use of trade secrets or proprietary information with respect to Longwood or any Contributed Entity.
- (e) No Contributed Entity maintains, operates, administers or contributes to, and no Contributed Entity is required to maintain, operate, administer or contribute to, any Benefit Plan. No Contributed Entity or any of its ERISA Affiliates, has, within the six years prior to the date hereof, sponsored, maintained, contributed to, or has ever had any obligation to contribute to any "pension plan" (as defined in Section 3(2) of ERISA) that is subject to Title IV of ERISA. Except with respect to intercompany arrangements with respect to the sharing of employee costs, no condition exists with respect to any Benefit Plan that could result in any Contributed Entity becoming liable directly or indirectly (by indemnification or otherwise) for any material liability under any such Benefit Plan, including by reason of affiliation of any Contributed Entity with an ERISA Affiliate thereof.
- (f) For purposes of this Agreement, "Material Benefit Plan" means each material Benefit Plan that is maintained, operated, administered, contributed to or required

to be contributed to by Longwood or any of its Affiliates for the benefit of any current or former director, manager, officer or employee of Longwood or any of its Affiliates who provides services to any Contributed Entity.

- (g) To the extent applicable to each Material Benefit Plan, (i) each Material Benefit Plan has been established, maintained and administered in all material respects in accordance with its terms and in compliance in all material respects with ERISA, the Code or other applicable Law, (ii) there are no material Legal Proceedings pending, or to the Knowledge of Longwood threatened, with respect to any Material Benefit Plan or any assets of any Material Benefit Plan (other than routine claims for benefits), and (iii) to the Knowledge of Longwood, no facts or circumstances exist that could reasonably be expected to give rise to such Legal Proceeding.
- (h) The consummation of the transactions contemplated hereby will not, either alone or in combination with another event, (i) entitle any current or former director or manager of any Contributed Entity or any current or former provider of services to any Contributed Entity to any severance pay, retention bonus, change in control payment or any other payment payable by any Contributed Entity or any of its Affiliates, (ii) accelerate the time of payment or vesting, or increase the amount of compensation due any such individual by any Contributed Entity or any of its Affiliates, or (iii) require any contributions or payments by any Contributed Entity or any of its Affiliates to fund any obligations under any Benefit Plan, or cause any Contributed Entity or any of its Affiliates to transfer or set aside any assets to fund any Benefit Plan.
- 4.13 <u>Litigation</u>. Except as set forth on <u>Schedule 4.13</u>, there are no Legal Proceedings pending or, to the Knowledge of Longwood, threatened (a) against any Contributed Entity or to which any Contributed Entity is otherwise a party or (b) against any Affiliate of a Contributed Entity and relating to the assets, properties or operations of any Contributed Entity, before any Governmental Body. No Contributed Entity is subject to any Order of any Governmental Body issued in Legal Proceedings in which (a) any Contributed Entity was a party or (b) any Affiliate of any Contributed Entity was a party and that relates to the assets, properties or operations of such Contributed Entity.

4.14 Compliance with Laws; Permits.

- (a) Other than with respect to Environmental Laws (which are addressed in <u>Section 4.15</u>) and except as set forth on <u>Schedule 4.14(a)</u>, each Contributed Entity is in compliance in all material respects with all Laws applicable to their respective businesses or the operation of the Midstream Assets. No Contributed Entity has received any written notice of or been charged with the material violation of any Laws.
- (b) The Contributed Entities hold all material Permits currently required for the ownership or operation of the Midstream Assets. To the Knowledge of Longwood, no Contributed Entity is in default or violation (and no event has occurred which, with notice or the lapse of time or both, would constitute a default or violation) in any material respect of any term, condition or provision of any such material Permit to which such Contributed Entity is a party.

- 4.15 Environmental Matters. The representations and warranties contained in Section 4.5, Section 4.6, Section 4.16 and this Section 4.15 are the sole and exclusive representations and warranties of Longwood pertaining or relating to any environmental, health or safety matters, including any arising under any Environmental Laws. Except as provided on Schedule 4.15:
 - (a) The Contributed Entities are and have been in compliance, in all material respects, with all applicable Environmental Laws and environmental provisions in any Real Property Leases, Rights-of-Way and Contracts, which compliance includes, but is not limited to, obtaining, maintaining and complying with all Permits required under all applicable Environmental Laws necessary to conduct the operation of the Midstream Assets;
 - (b) To the Knowledge of Longwood, there has not been any Release or threatened Release of any Hazardous Material, in either case, which would be reasonably likely to form the basis of any material Liability for any Contributed Entity;
 - (c) Neither Longwood nor any Contributed Entity has received any written notice or request for information and, to the Knowledge of Longwood, the Contributed Entities are not the subject of any investigation pending or threatened by any Governmental Body concerning or alleging a material violation of or liability under Environmental Law;
 - (d) No claims, actions, demands, suits or proceedings are pending or, to the Knowledge of Longwood, threatened against the Contributed Entities alleging a material violation of or Liability under Environmental Laws and the Contributed Entities are not subject to any material outstanding order, writ, judgment, award, injunction or decree of any Governmental Body or any arbitrator or arbitrators, in each case based on, arising out of, or relating to Environmental Law; and
 - (e) Longwood has provided all audits and other material reports, including all material Permits required under all applicable Environmental Laws, pertaining to the Contributed Entities' compliance with Environmental Law, and all "Phase I," "Phase II" or other environmental reports, in each case, in their possession or to which they have reasonable access regarding the present and past operations of the Contributed Entities.
- 4.16 <u>Insurance</u>. Except as set forth on <u>Schedule 4.16</u>, (a) the Midstream Assets and other assets and properties of the Contributed Entities are covered by insurance policies in full force and effect for such amounts as are sufficient for all requirements of Law and all agreements to which the Contributed Entities are a party or by which they or the Midstream Assets are bound, and (b) there are no outstanding claims under such policies relating to the Midstream Assets.
- 4.17 Related Party Transactions. Except as set forth on Schedule 4.17, there are no Contracts between (a) any Contributed Entity or any of its respective directors, managers, officers, employees or consultants and (b) Longwood or its Affiliates (which, for the purpose of this clause (b), does not include the Contributed Entities) or any of their respective directors, partners, managers, officers, employees or consultants, on the other hand, other than the Organizational Documents of the Contributed Entities.

- 4.18 <u>Gas Regulatory Matters</u>. No Contributed Entity is a "natural-gas company" under the Natural Gas Act of 1938 ("<u>NGA</u>"), and none of the Midstream Assets have ever been used in a manner that would require certification under the NGA or subject such assets to the jurisdiction of the Federal Energy Regulatory Commission. No Contributed Entity provides transportation or storage services pursuant to Section 311(a) of the Natural Gas Policy Act of 1978.
- 4.19 <u>Financial Advisors</u>. No Person has acted, directly or indirectly, as a broker, finder or financial advisor for Longwood, any of its Affiliates or any of the Contributed Entities in connection with the transactions contemplated by this Agreement who is entitled to any fee or commission or like payment for which Five Point, the Company or any Contributed Entity would be liable.
- 4.20 <u>Financing</u>; <u>Financial Capability</u>. Longwood's obligations hereunder are not subject to any conditions regarding its or any other Person's ability to obtain financing for the transactions contemplated hereby. Longwood has access to, and at the Closing will have, such funds as are necessary for the consummation by it of the transactions contemplated hereby.
- 4.21 <u>Investment Intent; Investment Experience</u>. Longwood is acquiring the Securities for investment only and not with a view toward, or for sale in connection with, any distribution thereof, nor with any intention of distributing or selling the Securities, in each case, in violation of the Securities Act or any other applicable Law. Longwood acknowledges and agrees that the Securities may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act, except pursuant to an exemption from such registration available under the Securities Act, and without compliance with any other applicable Law. Longwood acknowledges that it can bear the economic risk of its investment in the Securities, and has such knowledge and experience in financial and business matters and the industries in which the Company operates that it is capable of evaluating the merits and risks of an investment in the Securities.
- 4.22 <u>Solvency</u>. Longwood is not entering into the transactions contemplated by this Agreement with the intent to hinder, delay or defraud either present or future creditors.
- 4.23 <u>Independent Investigation</u>. Longwood has conducted to its satisfaction an independent investigation and verification of the current condition, assets and affairs of the Company and the Contributed Entities and the risks related thereto. Longwood further acknowledges that in making the decision to enter into this Agreement and to consummate the transactions contemplated hereby, Longwood has relied solely on (a) the results of such independent investigation and (b) the express written representations, warranties and covenants in this Agreement and the Longwood Agreements and has not, and will not, rely on any other statements, representations or advice from, or any financial models or projections prepared by, Five Point, the Company, any Contributed Entity or their respective representatives or Affiliates. Longwood acknowledges that it has had the opportunity to visit with Five Point and meet with its representatives to discuss the Company, the business of the Contributed Entities (including the Midstream Assets) and their conditions and prospects.

4.24 No Other Representations or Warranties; Schedules. Except for the representations and warranties contained in this Article IV (as modified by the Schedules hereto as supplemented or amended) and the Longwood Agreements, neither Longwood nor any other Contributed Entity nor any other Person makes any other express or implied representation or warranty with respect to Longwood or the Contributed Entities or the transactions contemplated by this Agreement, and Longwood disclaims any other representations or warranties, whether made by Longwood or any of its Affiliates, or any of their respective officers, directors, employees, agents or representatives. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS ARTICLE IV (AS MODIFIED BY THE SCHEDULES HERETO AS SUPPLEMENTED OR AMENDED) OR THE LONGWOOD AGREEMENTS, LONGWOOD HEREBY DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, PROJECTION, FORECAST, STATEMENT OR INFORMATION MADE, COMMUNICATED OR FURNISHED (ORALLY OR IN WRITING) TO FIVE POINT, THE COMPANY OR ANY OF THEIR AFFILIATES OR REPRESENTATIVES PRIOR TO THE CLOSING DATE (INCLUDING ANY OPINION, INFORMATION, PROJECTION OR ADVICE THAT MAY HAVE BEEN OR MAY BE PROVIDED TO FIVE POINT OR THE COMPANY BY ANY DIRECTOR, OFFICER, EMPLOYEE, AGENT, CONSULTANT OR REPRESENTATIVE OF LONGWOOD OR ANY OF ITS AFFILIATES). LONGWOOD MAKES NO REPRESENTATIONS OR WARRANTIES TO FIVE POINT OR THE COMPANY REGARDING THE PROBABLE SUCCESS OR PROFITABILITY OF THE BUSINESS OF THE COMPANY FOLLOWING THE CLOSING DATE.

ARTICLE V REPRESENTATIONS AND WARRANTIES OF FIVE POINT

Five Point hereby represents to Longwood and the Company that:

- 5.1 <u>Organization and Good Standing</u>. Five Point is a limited liability company duly organized, validly existing and in good standing under the laws of Delaware and has all requisite limited liability company power and authority to own, lease and operate its properties and to carry on its business.
- 5.2 <u>Authorization of Agreement</u>. Five Point has all requisite limited liability company power, authority and legal capacity to execute and deliver this Agreement and each other Transaction Agreement to which it is a party (together with this Agreement, the "<u>Five Point Agreements</u>"), and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and each of the Five Point Agreements and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all requisite limited liability company action on the part of Five Point. This Agreement has been, and each of the Five Point Agreements will be at or prior to the Closing, duly and validly executed and delivered by Five Point, and (assuming the due authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes, and each Five Point Agreement, when so executed and delivered will constitute, the legal, valid and binding obligation of Five Point, enforceable against Five Point in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity).

5.3 Conflicts; Consents of Third Parties.

- (a) None of the execution and delivery by Five Point of this Agreement or the Five Point Agreements, the consummation of the transactions contemplated hereby or thereby, or compliance by Five Point with any of the provisions hereof or thereof will conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination or cancellation under, any provision of (i) Five Point's organizational and governing documents; (ii) any Contract or Permit to which Five Point is a party or by which any of the properties or assets of Five Point are bound; (iii) any Order of any Governmental Body applicable to Five Point or by which any of the properties or assets of Five Point are bound; or (iv) any applicable Law, in each case, except as would not, individually or in the aggregate, adversely affect the ability of Five Point to consummate the transactions contemplated by this Agreement.
- (b) Except as set forth in Schedule 5.3(b), no consent, waiver, approval, Order, Permit or authorization of, or declaration or filing with, or notification to, any Person or Governmental Body is required on the part of Five Point in connection with the execution and delivery of this Agreement or the Five Point Agreements, or the compliance by Five Point with any of the provisions hereof or thereof, the consummation of the transactions contemplated hereby or the taking by Five Point of any other action contemplated hereby, except for the consents, waivers, approvals, Orders, permits or authorizations of, or declarations or filings with, any Governmental Body which (i) have been obtained or (ii) are customarily made or obtained following the Closing.
- 5.4 <u>Litigation</u>. There are no Legal Proceedings pending or, to the Knowledge of Five Point, threatened against Five Point or to which Five Point is otherwise a party before any Governmental Body or against any Affiliate of Five Point, in each case, (i) that would have a material effect upon the validity of this Agreement or any action taken or to be taken by Five Point in connection with, or which seek to enjoin or obtain monetary damages in respect of, this Agreement or (iii) that would reasonably be expected to adversely affect in any material respect the ability of Five Point to enter into this Agreement, perform its obligations under or consummate the transactions contemplated by this Agreement.
- 5.5 <u>Financial Advisors</u>. No Person has acted, directly or indirectly, as a broker, finder or financial advisor for Five Point in connection with the transactions contemplated by this Agreement who is entitled to any fee or commission or like payment for which Longwood, the Company or any Contributed Entity would be liable.
- 5.6 <u>Financing</u>; <u>Financial Capability</u>. Five Point's obligations hereunder are not subject to any conditions regarding its or any other Person's ability to obtain financing for the transactions contemplated hereby. Five Point has access to, and at the Closing will have, such funds as are necessary for the consummation by it of the transactions contemplated hereby.

- 5.7 <u>Investment Intent; Investment Experience</u>. Five Point is acquiring the Securities for investment only and not with a view toward, or for sale in connection with, any distribution thereof, nor with any intention of distributing or selling the Securities, in each case, in violation of the Securities Act or any other applicable Law. Five Point acknowledges and agrees that the Securities may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act, except pursuant to an exemption from such registration available under the Securities Act, and without compliance with any other applicable Law. Five Point acknowledges that it can bear the economic risk of its investment in the Securities, and has such knowledge and experience in financial and business matters and the industries in which the Company operates that it is capable of evaluating the merits and risks of an investment in the Securities.
- 5.8 <u>Independent Investigation</u>. Five Point has conducted to its satisfaction an independent investigation and verification of the current condition, assets and affairs of the Company and the Contributed Entities and the risks related thereto. Five Point further acknowledges that in making the decision to enter into this Agreement and to consummate the transactions contemplated hereby, Five Point has relied solely on (a) the results of such independent investigation and (b) the express written representations, warranties and covenants in the Transaction Agreements and has not, and will not, rely on any other statements, representations or advice from, or any financial models or projections prepared by, Longwood, the Company or their respective representatives or Affiliates. Five Point acknowledges that it has had the opportunity to visit with Longwood and meet with its respective representatives to discuss the Company, the business of the Contributed Entities (including the Midstream Assets) and their conditions and prospects.
- 5.9 No Other Representations or Warranties; Schedules. Except for the representations and warranties contained in this Article V or the Five Point Agreements, Five Point does not make any other express or implied representation or warranty with respect to Five Point or the transactions contemplated by this Agreement, and Five Point disclaims any other representations or warranties, whether made by Five Point or any of its Affiliates, or any of their respective officers, directors, employees, agents or representatives. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS ARTICLE V OR THE FIVE POINT AGREEMENTS, FIVE POINT HEREBY DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, PROJECTION, FORECAST, STATEMENT OR INFORMATION MADE, COMMUNICATED OR FURNISHED PRIOR TO THE CLOSING DATE (ORALLY OR IN WRITING) TO LONGWOOD, THE COMPANY OR THEIR AFFILIATES OR REPRESENTATIVES (INCLUDING ANY OPINION, INFORMATION, PROJECTION OR ADVICE THAT MAY HAVE BEEN OR MAY BE PROVIDED TO LONGWOOD, THE COMPANY OR THEIR AFFILIATES OR REPRESENTATIVES BY ANY DIRECTOR, OFFICER, EMPLOYEE, AGENT, CONSULTANT OR REPRESENTATIVE OF FIVE POINT OR ANY OF ITS AFFILIATES).

ARTICLE VI REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company hereby represents to Longwood and Five Point that:

- 6.1 <u>Organization and Good Standing</u>. The Company is a limited liability company duly formed, validly existing and in good standing under the laws of the State of Texas and has all requisite limited liability power and authority to own, lease and operate its properties and carry on its business.
- 6.2 <u>Authorization of Agreement</u>. The Company has all requisite limited liability company power and authority to execute and deliver this Agreement and each other Transaction Agreement to which it is a party (the "<u>Company Documents</u>"), and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by the Company of this Agreement and each Company Document have been duly authorized by all requisite limited liability company action on the part of the Company. This Agreement has been, and each Company Document will be at or prior to the Closing, duly executed and delivered by the Company and (assuming the due authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes, and each Company Document when so executed and delivered will constitute, the legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity).

6.3 Conflicts; Consents of Third Parties.

- (a) None of the execution and delivery by the Company of this Agreement or the Company Documents, the consummation of the transactions contemplated hereby or thereby or the compliance by the Company with any of the provisions hereof or thereof will conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination or cancellation under, any provision of (i) Organizational Documents of the Company; (ii) any Contract or Permit to which the Company is a party or by which the Company or its properties or assets are bound; (iii) any Order of any Governmental Body applicable to the Company or by which any of the properties or assets of the Company are bound; or (iv) any applicable Law, in each case, except as would not, individually or in the aggregate, adversely affect the ability of the Company to consummate the transactions contemplated by this Agreement.
- (b) No consent, waiver, approval, Order, Permit or authorization of, or declaration or filing with, or notification to, any Person or Governmental Body is required on the part of the Company in connection with the execution and delivery of this Agreement or the Company Documents, the compliance by the Company with any of the provisions hereof or thereof, the consummation of the transactions contemplated hereby or the taking by the Company of any other action contemplated hereby, except for the consents, waivers, approvals, Orders, permits or authorizations of, or declarations or filings with, any Governmental Body which (i) have been obtained or (ii) are customarily obtained following the Closing.
- 6.4 <u>Financial Advisors</u>. No Person has acted, directly or indirectly, as a broker, finder or financial advisor for the Company in connection with the transactions contemplated by this Agreement who is entitled to any fee or commission or like payment for which Longwood, any Contributed Entity, Five Point or the Company would be liable.

ARTICLE VII COVENANTS

7.1 <u>Further Assurances</u>. From time to time following the Closing, the Purchasers and the Company shall execute, acknowledge and deliver all such further conveyances, notices, assumptions, releases and acquittances and such other instruments, as reasonably necessary or appropriate to assure to the Parties, their Affiliates and their respective successors and permitted assigns, the benefits, rights and obligations of such Party under the Transaction Agreements, and to otherwise make effective the transactions contemplated hereby.

7.2 Tax Matters.

(a) Proration of Taxes. Longwood shall be responsible for, and shall indemnify the Company Indemnified Parties against any and all Damages to the extent arising from or relating to: (i) any and all Taxes (or the non-payment thereof) of the Contributed Entities attributable to (A) any Tax period ending on or prior to the Closing Date or (B) the portion of any Straddle Period ending on the Closing Date; (ii) any Taxes of any member of an affiliated, consolidated, combined or unitary group of which a Contributed Entity (or its predecessor) is or was a member on or prior to the Closing Date, including pursuant to Section 1.1502-6 of the Treasury Regulations or any analogous or similar state, local, or no-U.S. Law or regulation; (iii) any Taxes of any person (other than the Contributed Entities) imposed on a Contributed Entity as a transferee or successor or by contract which Taxes relate to an event or transaction occurring before the Closing; (iv) any Taxes arising out of or resulting from the breach or any of the representations or warranties in Section 4.7 (subject to Section 4.7(j)); and (v) any Taxes directly resulting from any breach or nonperformance by Longwood of any of the covenants contained in this Section 7.2; provided, however, that notwithstanding anything in this Section 7.2(a) or otherwise in this Agreement to the contrary, Longwood shall not be responsible for, nor indemnify the Company Indemnified Parties against, any Taxes that are taken into account in the calculation of the Final Net Working Capital. The Company shall be responsible for, and shall indemnify Longwood and its Affiliates against, all Taxes of the Company and any of its Subsidiaries (including the Contributed Entities) attributable to (A) any Tax period beginning after the Closing Date and (B) the portion of any Straddle Period beginning after the Closing Date, in each case, that are paid to a Tax authority by Longwood or its Affiliates (excluding the Company and its subsidiaries) (it being understood that Longwood shall generally bear its share of such Taxes indirectly as a Member of the Company following Closing Date). For the avoidance of doubt and notwithstanding anything else in this Agreement to the contrary, liability for (and any claims by any Company Indemnified Party or Longwood for indemnification of) any and all Damages arising from or relating to any Taxes described in Section 9.6 of the Company Agreement shall be governed solely by the agreement set forth in Section 9.6 of the Company Agreement.

- (b) For purposes of allocating any Taxes that are assessed for any Tax period that begins on or before and ends after the Closing Date (such period, a "<u>Straddle Period</u>"): (A) Taxes that are based upon or related to income or receipts or imposed on a transactional basis, shall be allocated to the portion of the Straddle Period in which the transaction giving rise to such Taxes occurred; *provided, however*, that deductions for depreciation or amortization, and any exemptions or allowances provided for under applicable Tax Laws that are a fixed dollar amount, or that are calculated or provided for on an annual basis, shall be apportioned on a time basis by assuming that an equal portion of such deductions, exemptions or allowances for the entire Straddle Period is allocable to each day in such Straddle Period, and (B) Taxes that are ad valorem, property or other Taxes imposed on a periodic basis pertaining to a Straddle Period shall be apportioned on a time basis by assuming that an equal portion of such ad valorem, property or other Taxes for the entire Straddle Period is allocable to each day in such Straddle Period.

 Longwood or the Company, as applicable, shall send to the other Party a statement that apportions each Tax pursuant to this Section 7.2(b) based upon the amount of Taxes actually invoiced and paid to the applicable Taxing Authority by such Party. Such statement shall be accompanied by proof of the Party's actual payment of such Taxes. Within 10 business days of receipt of each such statement and proof of payment, the non-paying Party shall reimburse the paying Party for the non-paying Party's allocated portion of such Taxes.
- (c) <u>Tax Refunds</u>. If the Company, the Contributed Entities, or any of their Affiliates receives a refund of Taxes for which Longwood is responsible pursuant to <u>Sections 7.2(a)</u>, or for which Longwood or Five Point is responsible pursuant to <u>7.2(e)</u>, the Company shall cause the amount of such Tax refund to be paid to Longwood or Five Point (as applicable) within 30 days following the receipt of such Tax refund. For the avoidance of doubt, the payment of such a Tax refund to Longwood or Five Point shall not be treated as a distribution for purposes of Sections 5.4, 5.5 or 7.3 of the Company Agreement.
- (d) <u>Amended Tax Returns</u>. Following the Closing, the Company will not, unless required under applicable Tax Law, file (or cause or permit any of the Contributed Entities to file) an amended Tax Return with respect to any of the Contributed Entities for a Tax period beginning before the Closing Date without the advance written consent of Longwood (such consent not to be unreasonably withheld, conditioned or delayed).
- (e) <u>Transfer Taxes</u>. All federal, state, local or foreign or other excise, sales, use, value added, transfer (including real property transfer or gains), stamp, documentary, filing, recordation and other similar Taxes that may be imposed or assessed in connection with the purchase and sale of the Securities as contemplated by this Agreement ("<u>Transfer Taxes</u>"), shall be borne 51% by Longwood and 49% by Five Point. The Parties shall cooperate in the preparation and filing of any Tax Returns for Transfer Taxes. Reasonable expenses incurred in connection with filing all necessary Tax Returns and other documentation with respect to all such Transfer Taxes shall be borne 51% by Longwood and 49% by Five Point.
- (f) <u>Tax Contests</u>. Longwood shall, at its sole expense, be entitled to participate in whole or in part, to employ counsel of its choice, and/or to take control of the

defense of any pending or threatened audits or other proceedings (including any resulting litigation), or assessments or adjustments, that relate to Taxes of the Company or any of the Contributed Entities for which Longwood may be liable hereunder ("<u>Tax Contests</u>"). If Longwood elects to assume the defense of a Tax Contest, Five Point shall be entitled to the participation rights set forth in <u>Section 9.4(b)</u> of the Company Agreement in respect of such Tax Contest, and Longwood shall not settle any Tax Contest without the consent of Five Point (which consent shall not be unreasonably withheld, delayed or conditioned).

- (g) <u>Tax Cooperation</u>. The Parties shall cooperate fully, as and to the extent reasonably requested by either Party, in connection with the filing of Tax Returns and any audit, litigation, or other proceeding with respect to Taxes relating to the Contributed Entities. Such cooperation shall include the retention and (upon another Party's request) the provision of records and information that are relevant to any such Tax Return or audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided under this Agreement. Longwood and the Company agree to retain all books and records with respect to Tax matters pertinent to the Contributed Entities relating to any taxable period beginning before the Closing Date until the expiration of the statute of limitations of the respective taxable periods and to abide by all record retention agreements entered into with any Taxing Authority.
- (h) <u>Allocation</u>. Within thirty (30) days after the determination of the Final Contribution Adjustment Amount pursuant to <u>Section 2.3</u>, Five Point shall provide Longwood with a proposed allocation of the agreed initial Book Basis of the Contributed Interests (\$350,000,000) (together with all other amounts properly taken into account under applicable Law) among the assets of the Contributed Entities as of the Closing (the "<u>Proposed Allocation</u>"). If, within thirty (30) days after the delivery of the Proposed Allocation, Longwood notifies Five Point in writing that Longwood objects to the allocation set forth in the Proposed Allocation, Five Point and Longwood shall endeavor to resolve such dispute within thirty (30) days in good faith. Any amounts remaining in dispute at the conclusion of such thirty (30) days period shall be submitted to the Independent Accountant for resolution in accordance with the terms and conditions of <u>Section 2.3(c)</u>, *mutatis mutandis*. The Proposed Allocation, as revised to reflect the resolution of any disputed items (the "<u>Final Allocation</u>"), shall be binding on Five Point, Longwood, and the Company and their respective Affiliates except as noted below. Neither Five Point, Longwood, or the Company, nor any of their respective Affiliates, shall take any position on any Tax Return that is inconsistent with the Final Allocation unless required to do so in good faith to settle a Tax dispute with a Taxing Authority (and in such case Five Point, Longwood, or the Company, as applicable, shall provide prompt notice that it (or its Affiliates) has taken such a position to the other parties).
- 7.3 <u>Confidentiality</u>. Except as provided in <u>Section 7.5</u> below, each Party hereby agrees that the confidentiality obligations contained in Section 12.2 of the Company Agreement shall apply to this Agreement, the Transaction Agreements and the transactions contemplated hereby as if (i) such confidentiality obligations were fully set forth herein and (ii) this Agreement, the Transaction Agreements and the transactions contemplated hereby and thereby were included in the definition of "Confidential Information" in the Company Agreement.

7.4 <u>Preservation of Records</u>. Notwithstanding anything in this Agreement to the contrary, Longwood may retain copies of any and all records and may retain the originals of those records related to Taxes and accounting and auditing matters, in each case, with respect to any period ending prior to the Closing Date, in which case Longwood shall instead provide the Company with copies of such retained records at Longwood's sole cost and expense; *provided*, that Longwood shall treat any such records as Confidential Information (as defined in the Company Agreement), and such records shall be subject to <u>Section 7.3</u>.

7.5 <u>Publicity</u>. None of Longwood, Five Point or the Company shall issue any press release or public announcement concerning this Agreement, the other Transaction Agreements or the consummation of the transactions contemplated hereby or thereby without obtaining the prior written approval of Longwood and Five Point, which approval will not be unreasonably withheld or delayed, unless, in the sole judgment of Longwood or Five Point, as applicable, disclosure is otherwise required by applicable Law or by the applicable rules of any stock exchange on which Affiliates of Longwood lists securities; *provided*, that, to the extent required by applicable Law, the Party intending to make such release shall use its commercially reasonable efforts consistent with such applicable Law to consult with the other Party with respect to the timing and content thereof.

7.6 Consents. Following the Closing, Longwood agrees that it will use commercially reasonable efforts to obtain, as promptly as practicable, each third-party consent or approval set forth on Schedule 7.6. Until each such third-party consent or approval is obtained, Longwood or Longwood's Affiliate, as applicable, will provide the Company with (and the Company assumes) the economic benefit, burdens, obligations and risks of ownership of the Contracts for which such consent or approval has not been obtained on or prior to the Closing (the "Retained Agreements"), and (A) Longwood will enforce, at the Company's sole cost and expense, any and all rights of the Company, Longwood or Longwood's Affiliate, as applicable, against third parties with respect to the Retained Agreements attributable to the period from and after the Closing, including instituting and prosecuting all proceedings against parties other than the Company, Longwood or Longwood's Affiliate, as applicable, which Longwood may in its reasonable discretion deem proper in order to assert or enforce any rights under the Retained Agreements or defending and compromising any and all actions, suits or proceedings in respect of the Retained Agreements, (B) the Company will be entitled to retain for its own account any amounts collected pursuant to the foregoing attributable to the period from and after the Closing, including any amounts payable as interest in respect thereof, (C) Longwood or Longwood's Affiliate, as applicable, will promptly pay to the Company when received all monies attributable to the period from and after the Closing received by Longwood or Longwood's Affiliate, as applicable, under the Retained Agreements, (D) the Company assumes and will perform for Longwood or Longwood's Affiliate, as applicable, at no cost to Longwood or Longwood's Affiliate, as applicable, all obligations to be performed under the terms of the Retained Agreements, except that, if the Retained Agreement is held in the name of Longwood or Longwood's Affiliate, Longwood

Longwood's Affiliate, as applicable, will continue to maintain in the name of Longwood or Longwood's Affiliate, as applicable, any required insurance under each of the Retained Agreements, (E) Longwood will not, and will cause its Affiliates not to, amend or modify, or assign or waive any rights under, any such Contract without the unanimous consent of the board of the Company, and (F) the Company will indemnify and hold harmless Longwood and its Affiliates from and against any Damages incurred by Longwood or its Affiliates under any Retained Agreement; *provided*, *however*, that the Company shall only be required to indemnify Longwood and its Affiliates under this Section 7.6 to the extent such Damages are based upon, result from or arise out of the Company's or its Affiliates' activities after the Closing.

7.7 <u>Additional Post-Closing Matters</u>. Following the Closing, Longwood agrees that it will (i) use commercially reasonable efforts to obtain, as promptly as practicable, each additional or amended right-of-way, third-party assignment or approval necessary to cure the matters set forth on <u>Schedule 4.8(c)</u> and (ii) deliver to Five Point documentation evidencing the foregoing.

ARTICLE VIII SURVIVAL; INDEMNIFICATION

8.1 <u>Survival</u>. The representations and warranties set forth in <u>Articles IV</u>, <u>V</u> and <u>VI</u> shall survive the Closing (a) indefinitely, with respect to the Fundamental Representations, (b) for a period of 12 months after the Closing, with respect to all other representations and warranties under this Agreement (other than representations and warranties contained in <u>Section 4.7</u> (Taxes)) and (c) until 60 days after the expiration of the applicable statute of limitations with respect to representations and warranties contained in <u>Section 4.7</u> (Taxes). The covenants and agreements of the Parties hereto contained in this Agreement which continue beyond the Closing shall survive the Closing until fully performed. Notwithstanding the preceding sentences, any breach of representation, warranty, covenant or agreement in respect of which indemnity may be sought under this Agreement shall survive the time at which it would otherwise terminate pursuant to the preceding sentences, if notice of the inaccuracy or breach thereof giving rise to such right of indemnity shall have properly been given (in accordance with this <u>Article VIII</u>) to the Party against whom such indemnity may be sought prior to the time at which it would otherwise terminate pursuant to the preceding sentences.

8.2 <u>Indemnification by the Purchasers</u>. Effective after the Closing, subject to the terms, conditions of and the limitations set forth in this <u>Article VIII</u>, each Purchaser hereby agrees to indemnify, severally but not jointly, the Company and its Affiliates and their respective equity holders, officers, directors, managers and employees ("<u>Company Indemnified Parties</u>") against and shall hold each of them harmless from any and all damage, liability, loss, fine, penalty, cost and expense (including court costs and reasonable consultants' and attorneys' fees and expenses) ("<u>Damages</u>") arising from, based upon, related to or associated with (i) any breach or inaccuracy of any of the representations and warranties of such Purchaser contained in <u>Article IV</u> or <u>Article V</u> of this Agreement, as applicable, (ii) any breach of any covenant or agreement of such Purchaser in this Agreement that survives Closing pursuant to this Agreement and (iii) solely with respect to Longwood, the matters specified in <u>Schedule 8.2</u> (the "<u>Specified Matters</u>"), EVEN IF SUCH DAMAGES ARE CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE (WHETHER SOLE, JOINT OR CONCURRENT), STRICT LIABILITY OR OTHER LEGAL FAULT OF ANY COMPANY INDEMNIFIED PARTY, INVITEE OR THIRD PERSON, AND WHETHER OR NOT CAUSED BY A PRE-EXISTING

CONDITION. Notwithstanding anything in this Agreement to the contrary, for purposes of this <u>Section 8.2</u>, (A) the amount of Damages in respect of any breach of a representation or warranty (excluding clause (b) of <u>Section 4.6</u> or any reference to "Material" in the term "Material Contract") shall be deemed to exist either if such representation or warranty is actually inaccurate or breached or would have been inaccurate or breached if such representation or warranty had not contained any limitation or qualification as to materiality or Material Adverse Effect (which instead will be read as any adverse effect or change), and (B) the amount of Damages in respect of any breach of a representation or warranty (including any deemed breach resulting from the application of clause (A) but excluding clause (b) of <u>Section 4.6</u> or any reference to "Material" in the term "Material Contract") shall be determined without regard to any limitation or qualification as to materiality or Material Adverse Effect (which instead will be read as any adverse effect or change) set forth in such representation or warranty.

8.3 Indemnification by the Company. Effective after the Closing, subject to the terms and conditions of this Article VIII, the Company hereby indemnifies the Purchasers and their respective Affiliates and their respective officers, directors, equity holders, managers and employees ("Purchaser **Indemnified Parties**") against and shall hold each of them harmless from any and all Damages arising from, based upon, related to or associated with (i) any breach or inaccuracy of any of the representations and warranties of the Company contained in Article VI of this Agreement and (ii) any breach of any covenant or agreement of the Company in this Agreement that survives Closing pursuant to this Agreement, in each case, EVEN IF SUCH DAMAGES ARE CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE (WHETHER SOLE, JOINT OR CONCURRENT), STRICT LIABILITY OR OTHER LEGAL FAULT OF ANY PURCHASER INDEMNIFIED PARTY, INVITEE, OR THIRD PERSON, AND WHETHER OR NOT CAUSED BY A PRE-EXISTING CONDITION; provided, however, that no Purchaser Indemnified Party shall be entitled to indemnification under this Section 8.3 with respect to any Damages (other than Damages relating to Taxes) in respect of the breach of any covenants or agreements of the Company the performance of which is in the control of such Purchaser. Notwithstanding anything in this Agreement to the contrary, for purposes of this Section 8.3, (A) the amount of Damages in respect of any breach of a representation or warranty shall be deemed to exist either if such representation or warranty is actually inaccurate or breached or would have been inaccurate or breached if such representation or warranty had not contained any limitation or qualification as to materiality or Material Adverse Effect (which instead will be read as any adverse effect or change), and (B) the amount of Damages in respect of any breach of a representation or warranty (including any deemed breach resulting from the application of clause (A)) shall be determined without regard to any limitation or qualification as to materiality or Material Adverse Effect (which instead will be read as any adverse effect or change) set forth in such representation or warranty.

8.4 Indemnification Claim Procedures.

(a) <u>Third Party Claims</u>. The Person seeking indemnification under this <u>Article VIII</u> (the "<u>Indemnified Party</u>") agrees to give prompt notice in writing to the Party against whom indemnity is to be sought (the "<u>Indemnifying Party</u>") of the assertion of any claim or the commencement of any suit, action or proceeding by any third party ("<u>Third-Party Claim</u>") in respect of which indemnity may be sought under this <u>Article VIII</u>. Such notice shall set forth in reasonable detail such Third-Party Claim and the basis

for indemnification (taking into account the information then available to the Indemnified Party), and the calculation of Damages. The Indemnified Party shall notify the Indemnifying Party within 10 days after it becomes aware of any Third-Party Claim in respect of which indemnity may be sought under this Article VIII; provided, that the failure to so notify the Indemnifying Party within such time period shall not relieve the Indemnifying Party of its obligations or liability hereunder, except to the extent such failure prejudices the Indemnifying Party's ability to defend the Third-Party Claim. Thereafter, the Indemnified Party shall deliver to the Indemnifying Party, promptly after the Indemnified Party's receipt thereof, copies of all notices and documents (including court papers) received by the Indemnified Party relating to the Third-Party Claim.

- (b) The Indemnifying Party shall be entitled to participate in the defense of any Third-Party Claim and, if it so chooses, to assume the defense thereof with counsel selected by the Indemnifying Party reasonably satisfactory to the Indemnified Party. Notwithstanding the foregoing, the Indemnifying Party shall continue to be entitled to assert any limitation of any claims contained in this Article VIII.
- (c) If the Indemnifying Party elects to assume the defense of any such Third-Party Claim, it shall notify the Indemnified Party in writing of its intent to do so and the Indemnifying Party shall not be liable to the Indemnified Party for legal expenses subsequently incurred by the Indemnified Party in connection with the defense thereof. The Indemnifying Party will have the right to assume control of such defense of the Third-Party Claim provided the Indemnifying Party shall keep the Indemnified Parties advised of the status of such Third-Party Claim and the defense thereof on a reasonably current basis. If the Indemnifying Party assumes the control of the defense of any Third-Party Claim in accordance with the provisions of this Section 8.4, the Indemnified Party shall be entitled to participate in the defense of any such Third-Party Claim and to employ, at its expense, separate counsel of its choice for such purpose, it being understood, however, that the Indemnifying Party shall continue to control such defense; provided, that, notwithstanding the foregoing, the Indemnifying Party shall pay the reasonable costs and expenses of such defense (including reasonable attorneys' fees and expenses) of the Indemnified Party if the Indemnified Party's outside counsel shall have reasonably concluded that (x) there are defenses available to the Indemnified Party that are different from or additional to those available to the Indemnifying Party, or (y) there is a conflict of interest that would make it inappropriate under applicable standards of professional conduct to have common counsel for the Indemnifying Party and the Indemnified Party. Notwithstanding anything to the contrary in this Agreement, the Indemnifying Party will not be permitted to settle, compromise or consent to the entry of a judgment with respect to which indemnification is being sought hereunder, in each case, without the Indemnified Party's prior written consent (such consent not to be unreasonably withheld, conditioned or delayed) unless the settlement, compromise or consent (i) includes an unconditional release of the Indemnified Party from all liability arising out of such Third-Party Claim and (ii) does not contain any admission or statement of any wrongdoing or liability on behalf of the Indemnified Party.
- (d) Each Party shall reasonably cooperate, and cause their respective Affiliates to reasonably cooperate, in the defense or prosecution of any Third-Party Claim and shall furnish or cause to be furnished such records, information and testimony, and attend such conferences, discovery proceedings, hearings, trials or appeals, as may be reasonably requested in connection therewith.

(e) <u>Direct Claims</u>. If the Indemnified Party has a claim for indemnity under this <u>Article VIII</u> against the Indemnifying Party that does not involve a Third-Party Claim, the Indemnified Party agrees to give prompt notice in writing of such claim to the Indemnifying Party. Such notice shall set forth in reasonable detail such claim and the basis for indemnification (taking into account the information then available to the Indemnified Party) and the calculation of Damages. The failure to promptly notify the Indemnifying Party shall not relieve the Indemnifying Party of its obligations hereunder, except to the extent such failure prejudices the Indemnifying Party's ability to defend the claim. If the Indemnifying Party does not notify the Indemnified Party within 45 calendar days following its receipt of such notice that the Indemnifying Party disputes its liability to the Indemnified Party under this <u>Article VIII</u>, such direct claim for indemnity specified by the Indemnified Party in such notice shall be conclusively deemed a liability of the Indemnifying Party under this <u>Article VIII</u> and the Indemnifying Party shall pay the amount of such liability to the Indemnified Party on demand, or in the case of any notice in which the amount of the direct claim for indemnity (or any portion thereof) is estimated, on such later date when the amount of such direct claim (or such portion thereof) becomes finally determined.

8.5 Limitations.

(a) <u>De Minimis</u>; <u>Deductible and Damages Cap</u>. No claim for indemnification may be brought by Company Indemnified Parties against Longwood unless the aggregate amount of Damages with respect to all such claims exceeds \$1,750,000 (the "<u>Deductible</u>"), at which point Company Indemnified Parties may recover all Damages in respect of claims against Longwood in excess of the Deductible. The Deductible shall not apply to (1) inaccuracies in or breaches of any of Longwood's Fundamental Representations, (2) the indemnification obligations set forth in clause (ii) of <u>Section 8.2</u>, including Longwood's indemnification obligations for any breach of any covenant or agreement of Longwood set forth in <u>Section 7.2</u>, or (3) the Specified Matters (collectively, the "<u>Limitation Exceptions</u>"). Damages subject to indemnification by Longwood pursuant to <u>Section 8.3(i)</u> (other than in respect of any claim with respect to a breach or inaccuracy of a Fundamental Representation) relating to any single breach or series of related breaches by Longwood shall not constitute Damages, and therefore shall not be applied toward the Deductible or be indemnifiable hereunder, unless such Damages relating to any such single breach or series of related breaches exceed \$150,000. The maximum aggregate liability for all indemnification claims by the Company Indemnified Parties (other than the Limitation Exceptions) against Longwood shall not exceed \$17,500,000 and the maximum aggregate liability for all indemnification claims by the Company Indemnified Parties under clause (i) of <u>Section 8.2</u> against either Purchaser shall not exceed \$175,000,000. The maximum aggregate liability for all indemnification claims by the Purchaser Indemnified Parties under clause (i) of Section 8.3 shall not exceed \$175,000,000.

- (b) <u>Consequential Damages</u>. No Indemnifying Party will have any Liability for any consequential, punitive, exemplary, special or indirect damages except to the extent that (i) in the case of any consequential damages, such damages are the natural, probable and reasonably foreseeable result of a breach of this Agreement or (ii) the payment of such damages is to an Indemnified Party in satisfaction of a Third-Party Claim.
- (c) <u>Sole Remedy</u>. The indemnification provided for in <u>Section 7.2</u>, <u>Section 8.2</u> and <u>Section 8.3</u> shall be the sole and exclusive remedy for damages available to any Indemnified Party for breach of any of the representations, warranties, covenants and agreements contained in this Agreement or any right, claim or action arising from this Agreement, other than actions for fraud or intentional misrepresentation; *provided*, *however*, that nothing contained in this Agreement shall in any manner limit a Party's right to seek specific performance, injunctive relief and other equitable relief to enforce any of the other Party's obligations under this Agreement.
- (d) <u>Mitigation</u>. Each Indemnified Party shall use commercially reasonable efforts to mitigate its Damages upon and after obtaining knowledge of any event or condition that would reasonably be expected to give rise to any Damages that are indemnifiable hereunder.
- (e) <u>Insurance and Other Amounts</u>. In calculating the amount of any Damages, the proceeds actually received by the Indemnified Party or any of its Affiliates under any insurance policy or pursuant to any claim, recovery, settlement or payment by or against any other Person, net of any costs or expenses incurred in connection with securing or obtaining such proceeds, shall be deducted.
- (f) <u>Reimbursement</u>. If an Indemnified Party recovers an amount in respect of any Damages that are the subject of indemnification hereunder after all or a portion of such Damages have been paid to such Indemnified Party by an Indemnifying Party pursuant to <u>Section 8.2</u> or <u>Section 8.3</u>, the Indemnified Party shall promptly remit to the Indemnifying Party the amount, if any, by which (i) the sum of (A) the amount paid by the Indemnifying Party to such Indemnified Party in respect of such Damages plus (B) the amount otherwise received in respect thereof, exceeds (ii) the full amount of such Damages.

ARTICLE IX MISCELLANEOUS

- 9.1 Expenses. Except as otherwise provided in this Agreement, each of Longwood and Five Point shall bear its own expenses incurred in connection with the negotiation and execution of this Agreement and each other Transaction Agreement and the consummation of the transactions contemplated hereby and thereby.
 - 9.2 Submission to Jurisdiction; Venue; Arbitration.
 - (a) The Parties irrevocably submit to the exclusive jurisdiction of any federal or state court located within Dallas, Texas over all claims or causes of action

(whether in contract, equity or tort) that may be based upon, arise out of or relate to this Agreement or any other document, certificate or instrument delivered pursuant hereto or the negotiation, execution, termination, performance or nonperformance of such documents (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or any other document, certificate or instrument delivered pursuant hereto or as an inducement to enter into this Agreement or otherwise arising from the relationship between the Parties), and each Party irrevocably agrees that all claims in respect of such dispute, claim or cause of action or any suit, action or proceeding related thereto may be heard and determined in such courts. The Parties irrevocably waive, to the fullest extent permitted by applicable Law, any objection which they may now or hereafter have to the laying of venue of any such dispute, claim or cause of action brought in such court or any defense of inconvenient forum for the maintenance of such dispute. Each of the Parties agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.

- (b) Other than as set forth in Section 2.3(c), any dispute or disagreement arising out of, or relating to, the formation, interpretation, performance, breach, enforcement or validity of this Agreement or any amendment or other modification hereto shall, unless the Parties otherwise agree, be submitted to and settled by arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association ("AAA"), now in effect, except to the extent modified herein. Any arbitration hearing shall be held in Dallas, Texas. Further, unless the Parties shall mutually agree otherwise and except as otherwise set forth below, the rules of evidence of the State of Texas shall govern the discovery and presentation of evidence in any arbitration hearing.
- (c) Within 30 days after submission of a dispute or disagreement to arbitration, each of Longwood and Five Point shall have the right to select one arbitrator. Within 15 days after the selection of an arbitrator by each of Longwood and Five Point, the two arbitrators selected shall select a third arbitrator. All arbitrators must be independent from each Party and its Affiliates. If either of Longwood or Five Point fails to appoint an arbitrator or an appointed arbitrator fails to appoint the third arbitrator within the prescribed 15 day period then, on reasonable notice to the other Party, either Longwood or Five Point may ask the AAA to appoint such arbitrators within 15 days of the request therefor with due regard for the selection criteria herein. The arbitrators selected pursuant to this Section 9.2 shall be qualified by education, experience or training to render a decision upon the issues of the dispute. Unless otherwise determined by the arbitrators in accordance with Section 9.2(d) below, each Party shall bear the costs incurred by such Party in connection with the procedures described in this Section 9.2 and the Company shall bear the fees and expenses of the arbitrators. Longwood and Five Point shall require the arbitrators to make a decision within 60 days after the appointment of the arbitrators; provided, however, that such time period shall be extended (but not beyond an additional 60 days) as reasonably necessary to complete discovery so long as the Party or Parties conducting such discovery are proceeding with reasonable diligence towards completion thereof, and the arbitrators' decision shall be rendered within 10 days of the hearing.

- (d) The decision of a majority of the arbitrators shall be final and binding upon the parties to the arbitration, and not subject to any appeal. The arbitration award may, at the discretion of the arbitrators, include an equitable allocation of the costs of the arbitration, including the fees of the arbitrators and the reasonable attorneys' fees of, and other expenses reasonably incurred during the arbitration by the parties, taking into account the merits (or lack thereof) of the parties' claims and defenses and the decisions of the arbitrators in respect thereof. Judgment on any arbitral award may be entered in any court in Dallas County, Texas having jurisdiction if the Party against whom the award is rendered does not comply therewith within 90 days after the date of the award, or by such other time as specified in the arbitral award. The parties shall keep the arbitration proceedings and the terms of any arbitration award confidential, unless filing for judgment on the arbitral award in any court having jurisdiction is made to enforce the arbitral award as permitted by this Section 9.2(d).
- (e) Notwithstanding anything herein to the contrary, in the event of (i) any claim or dispute between Longwood or any of its Affiliates, on the one hand, and the Company or any of its subsidiaries, on the other hand, or (ii) the exercise of any right of the Company or any of the Contributed Entities under any Transaction Agreement between the Company or any of the Contributed Entities, on the one hand, and a Member (as defined in the Company Agreement) or any of its Affiliates, on the other hand, the Member that is not a party to such Transaction Agreement (or an Affiliate of such Member) shall have the right to direct and control the Company's or such Contributed Entity's assertion of any such claim by or against the Company or such subsidiary and its resulting participation in the applicable arbitration proceedings under this Agreement or another Transaction Agreement and exercise the rights of the Company or such Contributed Entity under such Transaction Agreement.
- (f) The procedures specified in this <u>Section 9.2</u> shall be the sole and exclusive procedures for the resolution of disputes; *provided*, *however*, either Longwood or Five Point, without prejudice to the mandatory procedures of this <u>Section 9.2</u>, may file a complaint for purposes of tolling the statute of limitations or seek a preliminary injunction or other provisional judicial relief, if in its sole judgment such action is necessary to avoid irreparable damage or to preserve the status quo. Venue for such action shall be exclusively in Dallas County, Texas. Notwithstanding such action, the Members will continue to participate in good faith in the procedures specified in this <u>Section 9.2</u>.

9.3 Entire Agreement; Amendments and Waivers.

(a) The Transaction Agreements (and all Exhibits and Schedules hereto and thereto) contain the entire agreement and understanding between the Parties with respect to the subject matter hereof and thereof and supersede all prior agreements and understandings (including any offer letters or term sheets), whether written or oral, relating to such subject matter. The Parties have voluntarily agreed to define their rights, liabilities and obligations respecting the transactions contemplated by this Agreement exclusively in contract pursuant to the express terms and conditions of this Agreement, and the Parties expressly disclaim that they are owed any duties or entitled to any remedies not expressly set forth in this Agreement. Furthermore, the Parties each hereby acknowledge that this

Agreement embodies the justifiable expectations of sophisticated parties derived from arm's-length negotiations, and all Parties to this Agreement specifically acknowledge that no Party has any special relationship with another Party that would justify any expectation beyond that of an ordinary purchaser and an ordinary seller in an arm's-length transaction.

(b) The terms and provisions of this Agreement may be modified or amended only by a written instrument executed by each of the Parties, and compliance with any term or provision hereof may be waived only by a written instrument executed by each Party entitled to the benefits of the same. No action taken pursuant to this Agreement, including any investigation by or on behalf of any Party, shall be deemed to constitute a waiver by the Party taking such action of compliance with any representation, warranty, covenant or agreement contained herein. The waiver by any Party hereto of a breach of any provision of this Agreement shall not operate or be construed as a further or continuing waiver of such breach or as a waiver of any other or subsequent breach. No failure on the part of any Party to exercise, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of such right, power or remedy by such Party preclude any other or further exercise thereof or the exercise of any other right, power or remedy.

9.4 <u>Governing Law</u>. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS, WITHOUT GIVING EFFECT TO ANY CONFLICTS OF LAW PROVISIONS.

9.5 <u>Notices</u>. All notices and other communications under this Agreement shall be in writing and shall be deemed given by delivery in person or by overnight courier or by facsimile transmission to the Parties at the respective addresses set forth below (or at such other address as any Party shall have furnished to the others in accordance with the terms of this <u>Section 9.5</u>). All notices and other communications that are addressed as provided in or pursuant to this <u>Section 9.5</u> shall be deemed duly and validly given (a) if delivered in person or by overnight courier, upon delivery or (b) if delivered by facsimile transmission, upon transmission thereof and receipt thereof.

If to Longwood or the Company, to:

Longwood Midstream Holdings, LLC One Lincoln Centre 5400 LBJ Freeway, Suite 1500 Dallas, Texas 75240 Attention: General Counsel Facsimile: 972-371-5201

If to Five Point, to:

FP MMP Holdings LLC 825 Town and Country Lane, Suite 700 Attention: David N. Capobianco Facsimile: 713-351-0701 9.6 Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any Law or public policy, all other terms or provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

9.7 <u>Binding Effect; Assignment</u>. This Agreement shall be binding solely upon and inure to the benefit of the Parties and their respective successors and permitted assigns. Nothing in this Agreement shall create or be deemed to create any third party beneficiary rights in any person or entity not a Party to this Agreement except as expressly provided by <u>Section 9.8</u> and <u>Article VIII</u>. No assignment of this Agreement or of any rights or obligations hereunder may be made by any Party, directly or indirectly (by operation of law or otherwise), without the prior written consent of the other Parties and any attempted assignment without the required consents shall be void. No assignment of any obligations hereunder shall relieve the Parties of any such obligations. Upon any such permitted assignment, the references in this Agreement to the assigning Party shall also apply to any such assignee unless the context otherwise requires.

9.8 Non-Recourse. All claims or causes of action (whether in contract or in tort, in law or in equity) that may be based upon, arise out of or relate to this Agreement or any other document, certificate or instrument delivered pursuant hereto, or the negotiation, execution, performance or non-performance of this Agreement or any other document, certificate or instrument delivered pursuant hereto (including any representation or warranty made in or in connection with this Agreement or any other document, certificate or instrument delivered pursuant hereto or as an inducement to enter into this Agreement and the other documents delivered pursuant hereto have any shared or vicarious liability for the actions or omissions of any other Person (except to the extent set forth therein). No Person who is not a named Party to this Agreement or the other documents delivered pursuant hereto (or a successor or permitted assign to such parties), including without limitation any director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney or representative of any named Party to this Agreement ("Non-Party Affiliates"), shall have any Liability (whether in contract or in tort, in law or in equity, or based upon any theory that seeks to impose liability of an entity Party against its owners or affiliates) for any obligations or Liabilities arising under, in connection with or related to this Agreement or any other document, certificate or instrument delivered pursuant hereto (except to the extent set forth therein) or its negotiation or execution; and each Party hereto or thereto waives and releases all such Liabilities, claims and obligations against any such Non-Party Affiliates. The Parties acknowledge and agree that the Non-Party Affiliates are intended third-party beneficiaries of this Section 9.8.

9.9 <u>Counterparts</u>. This Agreement may be executed and delivered (including by electronic transmission (e.g., facsimile or pdf)) in one or more counterparts, all of which shall be considered one and the same agreement and shall become effective when one or more counterparts have been signed by each of the Parties and delivered to the other Parties, it being understood that all Parties need not sign the same counterpart.

** REMAINDER OF PAGE INTENTIONALLY LEFT BLANK **

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their respective authorized officers as of the date first written above.

PURCHASERS:

LONGWOOD MIDSTREAM HOLDINGS, LLC

By: /s/ Matthew V. Hairford

Name: Matthew V. Hairford

Title: President

FP MMP HOLDINGS LLC

By: /s/ David N. Capobianco

Name: David N. Capobianco

Title: Chief Executive Officer & Managing Partner

THE COMPANY:

SAN MATEO MIDSTREAM, LLC

By: /s/ G. Gregg Krug
Name: G. Gregg Krug
Title: Senior Vice President

Signature Page to Subscription and Contribution Agreement

MATADOR RESOURCES COMPANY

FOURTH SUPPLEMENTAL INDENTURE

FOURTH SUPPLEMENTAL INDENTURE (this "Supplemental Indenture"), dated as of February 17, 2017, among Longwood Midstream Holdings, LLC, a Texas limited liability Company (the "New Guarantor"), a subsidiary of Matador Resources Company, a Texas corporation (the "Company"), DLK Black River Midstream, LLC, a Texas limited liability company ("DLK Black River"), Black River Water Management Company, LLC, a Texas limited liability company ("Black River Water"), the other existing Guarantors (as defined in the Indenture referred to herein), the Company and Wells Fargo Bank, National Association, as trustee under the Indenture referred to herein (the "Trustee"). The New Guarantor and the existing Guarantors are sometimes referred to collectively herein as the "Guarantors," or individually as a "Guarantor."

WITNESSETH

WHEREAS, the Company and the existing Guarantors have heretofore executed and delivered to the Trustee an indenture, dated as of April 14, 2015, as supplemented and amended by the First Supplemental Indenture, dated as of October 1, 2015, the Second Supplemental Indenture, dated as of November 4, 2015 and the Third Supplemental Indenture, dated as of June 8, 2016 (as amended, the "*Indenture*"), relating to the 6.875% Senior Notes due 2023 (the "*Securities*") of the Company;

WHEREAS, DLK Black River and Black River Water (collectively, the "Contributed Entities") are parties to and Guarantors under the Indenture;

WHEREAS, <u>Section 4.9</u> of the Indenture in certain circumstances requires the Company to cause a Restricted Subsidiary (i) to become a Guarantor by executing a supplemental indenture and (ii) to deliver an Opinion of Counsel to the Trustee as provided in such Section;

WHEREAS, <u>Section 10.9</u> of the Indenture permits the release of the Subsidiary Guarantee of a Guarantor at such time as such Guarantor ceases to guarantee any other Indebtedness (as defined in the Indenture) of the Company or any other Guarantor under a Credit Facility (as defined in the Indenture); and

WHEREAS, pursuant to <u>Section 9.1</u> of the Indenture, the Company, the Guarantors and the Trustee are authorized to execute and deliver this Supplemental Indenture to amend or supplement the Indenture without the consent of any Holder;

NOW THEREFORE, to comply with the provisions of the Indenture and in consideration of the foregoing and for other good and valuable consideration, the receipt of which is hereby acknowledged, the New Guarantor, DLK Black River, Black River Water, the other Guarantors, the Company and the Trustee mutually covenant and agree for the equal and ratable benefit of the Holders of the Securities as follows:

1. CAPITALIZED TERMS. Capitalized terms used herein without definition shall have the meanings assigned to them in the Indenture.

- 2. AGREEMENT TO GUARANTEE. The New Guarantor hereby agrees, jointly and severally, with all other Guarantors, to unconditionally Guarantee to each Holder and to the Trustee the Obligations, to the extent set forth in the Indenture and subject to the provisions in the Indenture. The obligations of the Guarantors to the Holders of Securities and to the Trustee pursuant to the Subsidiary Guarantees and the Indenture are expressly set forth in Article X of the Indenture and reference is hereby made to the Indenture for the precise terms of the Subsidiary Guarantees.
- 3. RELEASE OF GUARANTOR. On the date hereof and effective upon the time that the Contributed Entities have ceased to guarantee any other indebtedness of the Company or any other Guarantor under a Credit Facility, the parties agree that each Contributed Entity is released as a party to and as a Guarantor under the Indenture and that each Contributed Entity has no further obligations or liabilities under its Subsidiary Guarantee or the provisions of the Indenture.
- 4. EXECUTION AND DELIVERY. The New Guarantor agrees that its Subsidiary Guarantee shall remain in full force and effect notwithstanding any failure to endorse on each Security a notation of such Subsidiary Guarantee.
- 5. NEW YORK LAW TO GOVERN. THE LAWS OF THE STATE OF NEW YORK SHALL GOVERN AND BE USED TO CONSTRUE AND ENFORCE THIS SUPPLEMENTAL INDENTURE.
- 6. COUNTERPARTS. The parties may sign any number of copies of this Supplemental Indenture. Each signed copy shall be an original, but all of them together represent the same agreement. This Supplemental Indenture may be executed in multiple counterparts which, when taken together, shall constitute one instrument. The exchange of copies of this Supplemental Indenture and of signatures by facsimile or PDF transmission shall constitute effective execution and delivery of this Supplemental Indenture as to the parties hereto and may be used in lieu of the original Supplemental Indenture for all purposes. Signatures of the parties hereto transmitted by facsimile or PDF shall be deemed to be their original signatures for all purposes.
 - 7. EFFECT OF HEADINGS. The Section headings herein are for convenience only and shall not affect the construction hereof.
- 8. THE TRUSTEE. Except as otherwise expressly provided herein, no duties, responsibilities or liabilities are assumed, or shall be construed to be assumed, by the Trustee by reason of this Supplemental Indenture. This Supplemental Indenture is executed and accepted by the Trustee subject to all the terms and conditions set forth in the Indenture with the same force and effect as if those terms and conditions were repeated at length herein and made applicable to the Trustee with respect hereto.

[signature page follows]

IN WITNESS WHEREOF, the parties hereto have caused this Supplemental Indenture to be duly executed and attested, all as of the date first above written.

MATADOR RESOURCES COMPANY

By: /s/ David E. Lancaster
Name: David E. Lancaster

Title: Executive Vice President

BLACK RIVER WATER MANAGEMENT COMPANY, LLC

By: /s/ David E. Lancaster

Name: David E. Lancaster
Title: Executive Vice President

DLK BLACK RIVER MIDSTREAM, LLC

By: /s/ David E. Lancaster
Name: David E. Lancaster
Title: Executive Vice President

LONGWOOD MIDSTREAM HOLDINGS, LLC

By: /s/ David E. Lancaster
Name: David E. Lancaster
Title: Executive Vice President

Signature Page to Fourth Supplemental Indenture

GUARANTORS:

DELAWARE WATER MANAGEMENT COMPANY, LLC LONGWOOD GATHERING AND DISPOSAL SYSTEMS GP, INC.

LONGWOOD MIDSTREAM SOUTH TEXAS, LLC LONGWOOD MIDSTREAM SOUTHEAST, LLC LONGWOOD MIDSTREAM DELAWARE, LLC

MATADOR PRODUCTION COMPANY

MRC ENERGY COMPANY

MRC DELAWARE RESOURCES, LLC

MRC ENERGY SOUTHEAST COMPANY, LLC

MRC ENERGY SOUTH TEXAS COMPANY, LLC

MRC PERMIAN COMPANY

MRC PERMIAN LKE COMPANY, LLC

MRC ROCKIES COMPANY

SOUTHEAST WATER MANAGEMENT COMPANY, LLC

By: /s/ David E. Lancaster
Name: David E. Lancaster
Title: Executive Vice President

LONGWOOD GATHERING AND DISPOSAL SYSTEMS, LP

By: Longwood Gathering and Disposal Systems GP, Inc., its general partner

By: /s/ David E. Lancaster
Name: David E. Lancaster
Title: Executive Vice President

Signature Page to Fourth Supplemental Indenture

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Trustee

/s/ John Stohlmann By:

Name: John Stohlmann Title: Vice President

Signature Page to Fourth Supplemental Indenture



NEWS RELEASE

MATADOR RESOURCES COMPANY ANNOUNCES FORMATION OF JOINT VENTURE FOR DELAWARE BASIN MIDSTREAM ASSETS

DALLAS, Texas, February 17, 2017 – Matador Resources Company (NYSE: MTDR) ("Matador" or the "Company"), today announced the formation of San Mateo Midstream, LLC ("San Mateo" or the "joint venture") a strategic joint venture between a wholly-owned subsidiary of Matador and a subsidiary of Five Point Capital Partners LLC ("Five Point") to operate and expand Matador's midstream assets in the Delaware Basin in Eddy County, New Mexico and Loving County, Texas (the "Midstream Assets"). Matador received \$171.5 million in connection with the formation of the joint venture and may earn up to an additional \$73.5 million in deferred performance incentives over the next five years. Matador will continue to operate the Midstream Assets and control the joint venture. At formation, Matador and Five Point owned 51% and 49% of the joint venture, respectively. The implied value of the Midstream Assets and the associated gathering, processing and disposal agreements entered into with Matador, as described below, was approximately \$500 million at closing after taking into account the performance incentives.

Five Point provided initial cash consideration of \$176.4 million to the joint venture in exchange for its 49% interest. Approximately \$171.5 million of this cash contribution by Five Point was distributed by the joint venture to Matador as a special distribution. Matador contributed the Midstream Assets and \$5.1 million in cash to the joint venture in exchange for its 51% interest. The parties to the joint venture also committed to spend up to an additional \$150 million in the aggregate to expand the joint venture's midstream operations and asset base.

San Mateo will continue to provide firm capacity service to Matador at market rates while also being a leading service provider to third party customers in and around Matador's Rustler Breaks and Wolf asset areas. San Mateo expects to expand the Black River Cryogenic Processing Plant in Matador's Rustler Breaks asset area from its current inlet capacity of 60 million cubic feet of natural gas per day to as much as 260 million cubic feet of natural gas per day. This expansion is expected to be operational as early as the first quarter of 2018 and serve both Matador and third party customers. San Mateo also plans to accelerate the buildout of oil, natural gas and water gathering lines throughout both the Rustler Breaks and Wolf asset areas, as well as to drill and complete at least one additional commercial salt water disposal well in the Rustler Breaks asset area in 2017.

Included in the Midstream Assets contributed by Matador to the joint venture are the following:

- The Black River Cryogenic Processing Plant in the Rustler Breaks asset area in Eddy County, New Mexico;
- One salt water disposal well and a related commercial salt water disposal facility in the Rustler Breaks asset area;

- Three salt water disposal wells and a related commercial salt water disposal facility in the Wolf asset area in Loving County, Texas; and
- All related oil, natural gas and water gathering systems and pipelines in both the Rustler Breaks and Wolf asset areas.

Matador retained its ownership in its midstream assets in South Texas and North Louisiana, which are not part of the joint venture.

In connection with the joint venture, Matador dedicated its current and future leasehold interests in the Rustler Breaks and Wolf asset areas pursuant to 15-year, fixed-fee natural gas, oil and salt water gathering agreements and salt water disposal agreements. In addition, Matador dedicated its current and future leasehold interests in the Rustler Breaks asset area pursuant to a 15-year, fixed fee natural gas processing agreement. The joint venture will provide Matador with firm service under each of these agreements in exchange for certain minimum volume commitments.

Matador also announced its plans to add a fifth operated drilling rig in the Delaware Basin beginning early in the second quarter of 2017. This rig will begin drilling in Matador's Rustler Breaks asset area. Matador anticipates operating the five drilling rigs in the Delaware Basin throughout the remainder of 2017, including three rigs primarily in its Rustler Breaks asset area, one rig primarily in its Wolf asset area and one rig primarily in its Arrowhead and Ranger asset areas. Matador intends to provide its 2017 operational and financial outlook and initial 2017 guidance with its fourth quarter and full-year 2016 earnings release scheduled for Wednesday, February 22, 2017.

Joseph Wm. Foran, Chairman and Chief Executive Officer of Matador, said, "We are very pleased and excited to announce this transaction with Five Point, which recognizes the significant value that Matador has already created with our midstream assets in the Delaware Basin. In addition, this joint venture gives us and our partners at Five Point the opportunity to continue to build out and expand these assets to further enhance their value for our respective stakeholders, while giving Matador the opportunity to retain operational control of these important assets, which provide critical support to our ongoing exploration and production operations in the Delaware Basin. San Mateo will not only provide midstream services for Matador but will also serve third party customers in and around our Rustler Breaks and Wolf asset areas. We had many opportunities to make a deal on these assets with a number of different companies over the past year, and we are confident that Five Point is the right joint venture partner for Matador. We look forward to working together closely with the Five Point team going forward.

"When we kicked off our midstream initiatives in the Delaware Basin in 2014, our primary goals were to ensure firm takeaway and processing capacity for our production and to generate significant value for our shareholders. Through the strong execution of our midstream team, combined with the success of our drilling efforts in both the Rustler Breaks and Wolf asset areas, we have achieved these goals. The Board and I congratulate our midstream team for the value it has created thus far, and we look forward to supporting this new joint venture as it continues to expand our midstream operations and create future value."

David Capobianco, CEO and Managing Partner of Five Point, said, "Our joint venture with Matador perfectly aligns with Five Point's strategy of proactively identifying industry leading management teams to partner with to build out world-class midstream infrastructure companies. The joint venture will build oil, NGL, gas and water infrastructure to support the needs of Matador, as our anchor customer, as well as third party producers in

the region who seek infrastructure solutions. We firmly believe that the joint venture will create significant value, as the Delaware is one of the most promising producing basins in North America, yet currently lacks sufficient permanent 'in-basin' midstream infrastructure."

About Matador Resources Company

Matador is an independent energy company engaged in the exploration, development, production and acquisition of oil and natural gas resources in the United States, with an emphasis on oil and natural gas shale and other unconventional plays. Its current operations are focused primarily on the oil and liquids-rich portion of the Wolfcamp and Bone Spring plays in the Delaware Basin in Southeast New Mexico and West Texas. Matador also operates in the Eagle Ford shale play in South Texas and the Haynesville shale and Cotton Valley plays in Northwest Louisiana and East Texas. Matador also conducts midstream operations in support of its exploration, development and production operations and provides natural gas processing, natural gas, oil and salt water gathering services and salt water disposal services to third parties on a limited basis.

For more information, visit Matador Resources Company at www.matadorresources.com.

About Five Point Capital Partners

Five Point Capital Partners is a private equity firm focused on midstream energy infrastructure and energy sector investments across North America. Five Point's investment strategy is to partner with, develop and support strong management teams through buyouts and growth capital investments within the midstream energy sector. Based in Houston, Texas, Five Point is currently investing from Five Point Capital Midstream Fund II L.P. and manages more than \$600 million of capital.

For more information, please visit www.fivepointcp.com.

Forward-Looking Statements

This press release includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1934, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. "Forward-looking statements" are statements related to future, not past, events. Forward-looking statements are based on current expectations and include any statement that does not directly relate to a current or historical fact. In this context, forward-looking statements often address expected future business and financial performance, and often contain words such as "could," "believe," "would," "anticipate," "intend," "estimate," "expect," "may," "should," "continue," "plan," "predict," "potential," "project," "hypothetical," "forecasted" and similar expressions that are intended to identify forward-looking statements, although not all forward-looking statements contain such identifying words. Actual results and future events could differ materially from those anticipated in such statements, and such forward-looking statements may not prove to be accurate. These forward-looking statements involve certain risks and uncertainties, including, but not limited to, the following risks related to financial and operational performance: general economic conditions; the Company's ability to execute its business plan, including whether its drilling program is successful; the joint venture's ability to expand the Black River Processing Plant, the timing of such expansion and the operating results thereof; the timing and operating results of the buildout by the joint venture of oil, natural gas and water gathering lines and the drilling of an additional salt water disposal well; changes in oil, natural gas and natural gas liquids; its ability to replace reserves and efficiently develop

current reserves; costs of operations; delays and other difficulties related to producing oil, natural gas and natural gas liquids; its ability to make acquisitions on economically acceptable terms; its ability to integrate acquisitions; availability of sufficient capital to execute its business plan, including from future cash flows, increases in its borrowing base and otherwise; weather and environmental conditions; and other important factors which could cause actual results to differ materially from those anticipated or implied in the forward-looking statements. For further discussions of risks and uncertainties, you should refer to Matador's filings with the Securities and Exchange Commission (the "SEC"), including the "Risk Factors" section of Matador's most recent Annual Report on Form 10-K and any subsequent Quarterly Reports on Form 10-Q. Matador undertakes no obligation and does not intend to update these forward-looking statements to reflect events or circumstances occurring after the date of this press release, except as required by law, including the securities laws of the United States and the rules and regulations of the SEC. You are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date of this press release. All forward-looking statements are qualified in their entirety by this cautionary statement.

Contact Information

Mac Schmitz Capital Markets Coordinator investors@matadorresources.com (972) 371-5225

Matador Resources Company and Subsidiaries

UNAUDITED PRO FORMA FINANCIAL INFORMATION

The following unaudited pro forma condensed consolidated financial information is presented to illustrate the effect on the historical financial position and operating results of Matador Resources Company (the "Company") of (a) the formation of San Mateo Midstream, LLC ("San Mateo Midstream"), a joint venture with FP MMP Holdings LLC, a subsidiary of Five Point Capital Partners LLC ("Five Point"), and the transactions contemplated by that certain Subscription and Contribution Agreement, dated as of February 17, 2017 (the "Contribution Agreement"), by and among Longwood Midstream Holdings, LLC, a wholly-owned subsidiary of the Company ("Longwood Holdings"), Five Point and San Mateo Midstream and (b) the Company's acquisition on February 14, 2017 of the remaining non-controlling interest in Fulcrum Delaware Water Resources, LLC ("Fulcrum Delaware Water Resources") not previously owned by the Company for approximately \$2.6 million ((a) and (b), collectively, the "Transactions"). Pursuant to the terms of the Contribution Agreement, (i) Longwood Holdings made an initial capital contribution to San Mateo Midstream consisting of all of the outstanding equity interests in certain subsidiaries of the Company that hold the Company's midstream assets in the Wolf and Rustler Breaks asset areas (the "Midstream Assets") and cash in the amount of \$5.1 million, in exchange for 51.0% of the membership interests in San Mateo Midstream and (ii) Five Point made an initial capital contribution to San Mateo Midstream of cash in the amount of \$176.4 million, of which \$171.5 million was immediately distributed to Longwood Holdings, partially as a reimbursement of capital expenditures by the Company and its subsidiaries with respect to the Midstream Assets, in exchange for 49.0% of the membership interests in San Mateo Midstream.

The unaudited pro forma condensed consolidated balance sheet as of September 30, 2016 is based on the historical financial statements of the Company as of September 30, 2016, after giving effect to the Transactions as if they had occurred on September 30, 2016. The unaudited pro forma condensed consolidated statements of operations for the nine months ended September 30, 2016 and the year ended December 31, 2015 are based on the historical financial statements of the Company for such periods after giving effect to the Transactions as if they had occurred on January 1, 2016 or January 1, 2015, respectively.

The unaudited pro forma condensed consolidated financial statements are presented for illustrative purposes only and do not purport to indicate the financial condition or results of operations of future periods or the financial condition or results of operations that actually would have been realized had the Transactions been consummated on the date or for the periods presented. The pro forma adjustments are based on available information and certain assumptions that the Company believes are reasonable as of the date hereof. Assumptions underlying the pro forma adjustments are described in the accompanying notes, which should be read in conjunction with the unaudited pro forma condensed consolidated financial statements. The unaudited pro forma condensed consolidated financial information also should be read in conjunction with the Company's historical consolidated financial statements and notes included in the Company's Annual Report on Form 10-K for the year ended December 31, 2015, filed on February 29, 2016, and Quarterly Report on Form 10-Q for the quarter ended September 30, 2016, filed on November 4, 2016.

Matador Resources Company and Subsidiaries

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET

(In thousands, except par value and share data)	As Reported	September 30, 2016 Adjustment	Pro-Forma	
ASSETS Current assets				
Cash	\$ 20,566	\$ 165,272 (1)	\$ 185,838	
Restricted cash	1,803	8,475 (2)	10,278	
Accounts receivable	1,005	0,475(2)	10,270	
Oil and natural gas revenues	27,739		27,739	
Joint interest billings	18,796		18,796	
Other	5,657	_	5,657	
Lease and well equipment inventory	3,182	<u>_</u>	3,182	
Prepaid expenses	3,277	<u></u>	3,277	
Total current assets	81,020	173,747	254,767	
Property and equipment, at cost	01,020	175,747	254,707	
Oil and natural gas properties, full-cost method				
Evaluated	2,341,342	<u>_</u>	2,341,342	
Unproved and unevaluated	445,421	<u>_</u>	445,421	
Other property and equipment	141,420	<u></u>	141,420	
Less accumulated depletion, depreciation and amortization	(1,832,478)	_	(1,832,478)	
Net property and equipment	1,095,705		1,095,705	
Other assets	968	<u>—</u>	968	
		e 172 747		
Total assets	\$ 1,177,693	<u>\$ 173,747</u>	\$ 1,351,440	
LIABILITIES AND SHAREHOLDERS' EQUITY				
Current liabilities				
Accounts payable	\$ 4,534	\$ —	\$ 4,534	
Accrued liabilities	93,339	_	93,339	
Royalties payable	21,717	_	21,717	
Amounts due to affiliates	7,033	_	7,033	
Derivative instruments	10,139	_	10,139	
Advances from joint interest owners	3,847	_	3,847	
Deferred gain on plant sale	6,440	_	6,440	
Amounts due to joint ventures	4,050	_	4,050	
Other current liabilities	530		530	
Total current liabilities	151,629	_	151,629	
Long-term liabilities				
Borrowings under Credit Agreement	65,000	_	65,000	
Senior unsecured notes payable	392,153	_	392,153	
Asset retirement obligations	19,452	_	19,452	
Amounts due to joint ventures	2,700	_	2,700	
Derivative instruments	3,838	_	3,838	
Deferred gain on plant sale	97,676	_	97,676	
Other long-term liabilities	7,451		7,451	
Total long-term liabilities	588,270	_	588,270	
Shareholders' equity				
Common stock - \$0.01 par value, 120,000,000 shares authorized; 93,580,969 shares issued and			222	
93,464,898 shares outstanding	936		936	
Additional paid-in capital	1,176,198	134,991 (3)	1,311,189	
Accumulated deficit	(740,505)		(740,505)	
Total Matador Resources Company shareholders' equity	436,629	134,991	571,620	
Non-controlling interest in subsidiaries	1,165	38,756 (4)	39,921	
Total shareholders' equity Total liabilities and shareholders' equity	437,794 \$ 1,177,693	173,747 \$ 173,747	\$ 1,351,440	

- (1) Represents \$176.4 million of cash contributed by Five Point in connection with the formation of San Mateo Midstream less (i) approximately \$2.6 million paid to acquire the non-controlling interest in Fulcrum Delaware Water Resources not previously owned by the Company and (ii) \$10.0 million of cash restricted to operations of San Mateo Midstream. Also reflects \$1.5 million released from restriction upon the purchase of the non-controlling interest in Fulcrum Delaware Water Resources not previously owned by the Company.
- (2) Represents \$10.0 million in cash contributed to San Mateo Midstream less \$1.5 million released from restriction upon the purchase of the non-controlling interest in Fulcrum Delaware Water Resources not previously owned by the Company.
- (3) Reflects the purchase of the non-controlling interest in Fulcrum Delaware Water Resources not previously owned by the Company and the amount received in connection with the formation of San Mateo Midstream.
- (4) Represents the adjustment required to reflect the purchase of the non-controlling interest in Fulcrum Delaware Water Resources not previously owned by the Company and Five Point's 49% non-controlling interest in San Mateo Midstream.

Matador Resources Company and Subsidiaries

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(In thousands, except per share data)		Nine Months Ended September 30, 2016			Year Ended December 31, 2015	
	As Reported	Adjustment	Pro-Forma	As Reported	Adjustment	Pro-Forma
Revenues			· <u> </u>			
Oil and natural gas revenues	\$ 196,341	\$ —	\$ 196,341	\$ 278,340	\$ —	\$ 278,340
Third-party midstream services revenues	2,956	_	2,956	1,864	_	1,864
Realized gain on derivatives	10,413	_	10,413	77,094	_	77,094
Unrealized loss on derivatives	(30,261)		(30,261)	(39,265)		(39,265)
Total revenues	179,449	_	179,449	318,033	_	318,033
Expenses						
Production taxes, transportation and processing	30,846	_	30,846	35,650	_	35,650
Lease operating	41,300	_	41,300	54,704	_	54,704
Plant and other midstream services operating	3,537	_	3,537	3,489	_	3,489
Depletion, depreciation and amortization	90,185	_	90,185	178,847	_	178,847
Accretion of asset retirement obligations	828	_	828	734	_	734
Full-cost ceiling impairment	158,633	_	158,633	801,166	_	801,166
General and administrative	39,506		39,506	50,105		50,105
Total expenses	364,835	_	364,835	1,124,695	_	1,124,695
Operating income (loss)	(185,386)		(185,386)	(806,662)		(806,662)
Other income (expense)						
Net gain on asset sales and inventory impairment	3,140	_	3,140	908	_	908
Interest expense	(20,244)	_	(20,244)	(21,754)	_	(21,754)
Other (expense) income	(17)	_	(17)	616	_	616
Total other expense	(17,121)		(17,121)	(20,230)		(20,230)
Income (loss) before income taxes	(202,507)		(202,507)	(826,892)		(826,892)
Income tax (benefit) provision	(- , ,		(-))	(= -,)		(1 1,11)
Current	(1,141)	_	(1,141)	2,959	_	2,959
Deferred		_		(150,327)	_	(150,327)
Total income tax benefit	(1,141)		(1,141)	(147,368)		(147,368)
Net income (loss)	(201,366)		(201,366)	(679,524)		(679,524)
Net income attributable to non-controlling interest in	(201,500)		(201,500)	(073,324)		(0/3,324)
subsidiaries	(209)	(2,290)(1)	(2,499)	(261)	(1,388) (1)	(1,649)
Net loss attributable to Matador Resources	# (DO4 EEE)	d (2.200) (1)	# (DOD DOE)	φ (6 50 505)	d (4 DOO) (1)	ф. (CO4. 4 ED)
Company shareholders	<u>\$(201,575)</u>	\$ (2,290) (1)	\$(203,865)	\$ (679,785)	\$ (1,388) (1)	\$ (681,173)
Earnings (loss) per common share						
Basic	\$ (2.24)	\$ (0.02)	\$ (2.26)	\$ (8.34)	\$ (0.01)	\$ (8.35)
Diluted	\$ (2.24)	\$ (0.02)	\$ (2.26)	\$ (8.34)	\$ (0.01)	\$ (8.35)
Weighted average common shares outstanding						
Basic	90,016		90,016	81,537		81,537
Diluted	90,016		90,016	81,537		81,537

⁽¹⁾ Represents Five Point's interest in the net income of San Mateo Midstream during the periods.